In support of an OT-DM model: Evidence from a morphological conspiracy in Degema

This paper provides support for a modified DM model which I call Optimality Theoretic Distributed Morphology (OT-DM). The strongest form of this model is that all morphological operations take place in parallel, which I call the Morphology-in-Parallel Hypothesis (MPH). Although combining OT and DM is unorthodox in practice, I show that a growing body warrants his modification (Trommer 2001a, 2001b, 2002; Dawson *in press*; Foley *to appear*; a.o.), and provide evidence for OT-DM from a morphological conspiracy involving verbal clitics in Degema (Nigeria). To account for this conspiracy, I argue that agreement clitics are inserted post-syntactically via the DM operation Dissociated Node Insertion (DNI), and further that verb complexes are formed post-syntactically via the operation Local Dislocation (LD). I argue that both these operations are triggered by a well-formedness markedness constraint which requires verbs to appear in properly inflected words on the surface. These DM operations are subject to all DM operations, and are evaluated via EVAL against the ranked constraint set. I illustrate that under standard views of DM in which DNI proceeds VI, a serial model results in an ordering paradox, and that even after parameterizing DM operation ordering in response, this model does not adequately account for the morphological conspiracy.

distributed morphology optimality theory morphological conspiracy clitics serial verb constructions African linguistics

1. Introduction

This paper seeks to contribute to refining a generative model of the interface between syntax and phonology/phonetics. As a starting point, I adopt the major tenets of Distributed Morphology (DM) (Halle & Marantz 1993), one of which being that the syntactic module generates a structure which is then manipulated in a morphology module post-syntactically via a number of operations. These operations include vocabulary insertion, linearization, local dislocation, prosodic inversion, impoverishment, fusion, fission, dissociated node insertion, among others. The orthodox view of DM is that it has a localist rather than globalist architecture (Embick 2010), within which these operations take place serially in a specific order (e.g. Arregi & Nevins 2012).

In contrast, this paper maintains the core assumptions of DM but implements it within a constraint-based Optimality Theory (OT) model, which I call **Optimality Theoretic Distributed Morphology (OT-DM)**. The strongest form of this model is that all morphological operations take place in parallel, which I call the **Morphology-in-Parallel Hypothesis (MPH)**. Although combining OT and DM is unorthodox in practice, I show that a growing body of work overtly or covertly adopts it (Trommer 2001a, 2001b, 2002; Dawson *in press*; Foley *to appear*; a.o.), and illustrate that an OT-DM model is necessary to account for a **morphological conspiracy** in Degema, a language of southern Nigeria. Under OT-DM, syntax can be understood as generative and contains only syntactic operations (e.g. MERGE) while post-syntax is strictly evaluative and consists of familiar OT components (EVAL, GEN, CON, etc.).

The core data from Degema I introduce to support this position involve the distribution of verbal clitics in serial verb constructions. All verbs are marked with a proclitic expressing subject agreement and (when applicable) an enclitic expressing aspect, a cl=[V]=cl structure. In serial verb constructions, when the verbs are adjacent or when only a prosodically light pronoun intervenes, the proclitic appears on the first verb and the enclitic appears on the last verb. I refer to this as the **single-marking clitic pattern**. In contrast, when the verbs are separated by a prosodically heavy constituent, each verb is marked by a proclitic and enclitic. I refer to this as the **double-marking pattern**. These patterns' complementary distribution is shown below.

V ₁ V ₂	Double-marking	Single-marking
Ø	* $cl=[V_1]=cl cl=[V_2]=cl$	$\sqrt{cl=[V_1 V_2]=cl}$
Pron _σ	* $cl=[V_1 \mathbf{D}_{\sigma}]=cl cl=[V_2]=cl$	$\sqrt{cl=[V_1 \mathbf{D}_{\sigma} V_2]=cl}$
DP/Pron _{oo}	$\sqrt{cl=[V_1]=cl \mathbf{DP}/\mathbf{D}_{\sigma\sigma} cl=[V_2]=cl}$	* $cl=[V_1 \mathbf{DP}/\mathbf{D}_{\sigma\sigma} V_2]=cl$

Table 1: Complementary distribution of clitic marking in SVCs

I analyze certain of these clitics as agreement nodes which are inserted post-syntactically via the DM operation **Dissociated Node Insertion (DNI)**. Further, I analyze the verb complex in single-marking structures as a single morphological word which is formed post-syntactically via the operation **Local Dislocation (LD)**. I argue that both these operations are triggered by a well-formedness constraint which requires verbs to appear in properly inflected words on the surface, and can therefore by understood as forming a conspiracy in the sense of Kisseberth (1970). Both of these operations apply in parallel. Informally, if the verbs are sufficiently local defined prosodically, they may form a constituent to 'share' the clitic marking resulting in single-marking via LD; if however they are not sufficiently local, each verb receives its own set resulting in double-marking.

I implement this within the OT-DM model. The DM operations are decomposed into a series of constraints which are crucially ranked, e.g. LD is decomposed into constraints Markedness constraints (e.g. MWD=PRWD), Alignment constraints (e.g. ALIGN-/asp/-R), and Mapping constraints (e.g. LinearityMap-IO, MAP(WD_TYPE)). Candidates are freely generated from GEN and are subject to all DM operations, and evaluated via EVAL against the ranked constraint set. Under OT-DM, DM operations are freely licensed but only those candidates which are selected as optimal surface and demonstrate their effects. I illustrate how this model correctly generates the Degema clitic patterns, and adequately captures the morphological conspiracy. I contrast this model with three inadequate alternatives: a serial rule-based DM account, an account of single-marking via syntactic verb movement, and an account of single-marking via deletion-under-identity of intermediate clitics. With respect to the first alternative, I illustrate that under standard views of DM in which DNI proceeds VI, a serial model results in an **ordering paradox**, and that even after parameterizing DM operation ordering in response, this model does not adequately account for the morphological conspiracy.

This paper is organized as follows. Section 2 presents the major tenets of DM, and compares two different DM models, Rules & Constraints DM vs. OT-DM. Section 3 presents the core data involving clitics in Degema serial verb constructions, as well as core assumptions of the syntax of these structures. Section 4 presents a first pass DM analysis, introducing post-syntactic DM operations (DNI and LD), illustrating how evidence from grammatical tone supports the constituency of V+V verb clusters under adjacency, and discusses core problems in assuming a standard serial rule-based DM model of this data. Section 5 presents the core of the OT-DM analysis, introducing the constraint set, the candidate set, and the evaluation. Finally, section 6 presents arguments against two competing alternative accounts of this data, section 7 discusses the ramifications of OT-DM for post-syntactic architecture, and section 8 provides the conclusion. Two appendices are included which further detail the OT model.

2. Distributed Morphology, conspiracies, and constraints

2.1. Context of DM

One prominent theory which has emerged in parallel with Minimalism is **Distributed Morphology (DM)** (Halle & Marantz 1993, 1994; Halle 1997; Marantz 1997; Harley & Noyer 1999; Embick & Noyer 1999, 2001, 2007; Embick 2007a, 2007b, 2010; Embick & Marantz 2008; Siddiqi 2009, 2010; Arregi & Nevins 2012; Matushansky & Marantz 2013; Bobaljik 2017; a.o.). The core analytic move of DM is that morphology is distributed across several components of Grammar. Traditional 'morphemes' are decomposed into morphosyntactic feature bundles, i.e. roots and functional heads from the **Feature Lexicon** (*=narrow lexicon* – Marantz 1997:204), which are manipulated by standard syntactic operations. Other components of the traditional 'morpheme' are filled in post-syntactically in a later module. On the PF branch, **Vocabulary Items** from the **Vocabulary** insert phonological information; on the LF branch, idiosyncratic meaning comes from the **Encyclopedia**.

Conceptually, DM's theoretical commitments can be distinguished by a number of tenets along several logically separate axes.

(1) Tenets of DM

- a. Module Order: Syntax precedes morphology
- b. Internal Complexity: Internally complex words are the concatenation of morphosyntactic feature bundles
- c. Uniform Concatenation: Morphologically complex words are formed via the same operations concatenating words in clauses
- d. Phonology-Free: Syntax/semantics lack phonological features
- e. Feature Realization: Vocabulary Items realize features already present
- f. Bundle Manipulation: The output of feature bundles from syntax is manipulated via morphological operations
- g. Derivation Architecture: Input-to-output mapping is serial and rule-based

The first tenet Module Order is that syntax precedes 'morphology', which itself is decomposed into one or more post-syntactic modules. Morphology is not a module which takes place before or parallel to syntax as a whole. Second, Internal Complexity states that internally complex words result from concatenation of morphosyntactic feature bundles. This is in contrast with Inferential models of morphology which posit no word-internal structure and derive morphologically complex words by root/stem manipulation, e.g. Anderson's (1992) A-Morphous Morphology and Stump's (2001) Paradigm Function Morphology (see discussion in Inkelas 2014:241). Third, Uniform Concatenation states that morphologically complex words are formed via the same operations concatenating words in clauses, namely MERGE. This is therefore a strictly Non-Lexicalist theory of morphology, in which all words regardless of internal complexity have 'syntax-all-the-way-down' (Bobaljik 2017); 'wordhood' here is largely epiphenomenal. This contrasts with Lexicalist theories where there is a separate computational system which builds words independent from operations in syntax. Fourth, syntax and semantics are Phonology-Free and lack the presence of and sensitivity to phonological features ("nodes consist entirely of morphosyntactic/semantic features and lack phonological features" - Halle & Marantz 1993:121). Fifth, Feature Realization refers to the fact that vocabulary items (VIs) expone syntactic feature bundles and thereby realize syntactic features rather than introduce them (Late Insertion in DM terminology). This Realizational model contrasts with an Incremental model, in which the morphemes themselves introduce relevant syntactic/semantic features (Stump 2001). Sixth, **Bundle Manipulation** refers to the output of syntax - which consists of bundles of morphosyntactic features – able to be manipulated via morphological operations (or 'adjustments'), e.g. adding or deleting features. Finally, Derivation Architecture refers to the fact that in standard DM, post-syntactic operations apply serially, formalized as ordered rules which can feed or bleed one another. It is only this last tenet that the OT-DM model challenges and is the focus for the rest of the paper.

2.2. Morphological conspiracies

In rule-based phonological theory, a **conspiracy** refers to situations where rules with dissimilar inputs, outputs, and/or environments result in surface forms which all converge on or avoid a specific structure. The challenge of conspiracies for phonological theory was most famously presented in Kisseberth (1970) using data from Yokuts [yok] (=Yowlumne/Yawelmani; California, USA – Newman 1944). Yokuts has a number of rules involving vowel insertion, vowel deletion, and consonant deletion. In the simplified sample of rules is in (2), rules *i-Epenthesis* and *C-Deletion* have distinct inputs, outputs, and environments, but they both can be understood as avoiding a tautosyllabic consonant cluster, i.e. a 'derivational constraint' on surface structures, a shorthand for which is *CC.

- (2) Rules in the Yokuts phonological conspiracy avoiding *CC
 - a. i-Epenthesis: $\emptyset \rightarrow i / C_CC$
 - b. C-Deletion: $C \rightarrow \emptyset / \overline{CC} +$
 - c. Final V-Deletion: $V \rightarrow \emptyset / VC =$

Further, the non-application of a rule also conspires to avoid this constraint. *Final V-Deletion* in c. deletes word-final vowels if it is preceded by a VC sequence, but not if it is preceded by a CC sequence as this would create a banned surface structure. Conspiracies can also be analyzed as operating parametrically across languages, e.g. the cross-linguistic dispreference for vowel hiatus has a variety of repairs, e.g. vowel deletion, coalescence, gliding, and glide insertion among them. All of these rules can be understood as avoiding a sequence *VV.

Such conspiracies acted as catalysts in the development of parallelist constraint-based evaluation in **Optimality Theory (OT)**, given that such conspiracies are in fact predicted from a constraint-based model and are not readily captured in a rule-based one (Prince & Smolensky 2004 [1993]; Kager 1999: Sec. 2.1.1.2; McCarthy 2002:Sec.3.1.4.3, 2008:1-12; Bermúdez-Otero & Börjars 2006:720; Kisseberth 2011; a.o.). A **morphological conspiracy** therefore is surface structure convergence or avoidance involving morphological (i.e. non-phonological) inputs, outputs, and environments. Unlike in phonological theory, however, morphological conspiracies have played virtually no role in the major developments of morphological theory. Despite their minor role, several morphological conspiracies have been identified in the literature. I highlight here two very different responses to conspiracies within DM theory, one maintaining ordered rules and the other proposing strictly parallel constraints.

2.3. Rules & Constraints DM

One response to morphological conspiracies is exemplified by Arregi & Nevins' (2012) thorough investigation of Basque morphosyntax. They posit a series of ordered morphological modules in a rule-based DM model, summarized below.

- (3) Ordered morphological modules (Arregi & Nevins 2012:4)
- [...syntactic operations...]
- 1. Exponence Conversion (e.g. agree-copy, fission)
- 2. Feature Markedness (e.g. participant dissimilation)
- 3. Morphological Concord (e.g. complementizer agreement)
- 4. Linearization
- 5. Linear Operations (e.g. clitic metathesis/doubling)
- 6. Vocabulary Insertion
- [...phonological operations...]

Arregi & Nevins highlight a number of surface patterns across Basque dialects which conspire to place T° auxiliaries in a specific position. One such conspiracy is avoiding an auxiliary being initial in its relevant domain. Several processes 'conspire to make T surface in second position', which include cliticization of an absolutive marker (*Absolutive Cliticization*), metathesis and doubling of an ergative marker whose application depends on the dialect (*Ergative Metathesis and Doubling*), and the insertion of an epenthetic dummy morpheme before T^o (*L*-*Support*).

To unify these distinct repairs, Arregi & Nevins posit a well-formedness constraint T-NONINITIALITY (Arregi & Nevins 2012: 276). This constraint acts as an inviolable filter that outputs must comply with, and triggers 'repairs' which vary by environment/module and by dialect. By employing both rules and constraints, Arregi & Nevins propose a significant modification to the standard DM model, which I refer to as **Rules & Constraints Distributed Morphology (R&C DM)**. Their modification to DM's rule-based architecture in response to output-oriented well-formedness conditions is directly parallel to the events in pre-OT phonology, such as Paradis' (1987) *Constraints and Repair Strategies* (for further discussion see, Prince & Smolensky 2004 [1993]: 247-252 and Kiparsky 2017:396).

It is crucial to note that Rules & Constraints DM explicitly rejects a more radical OT model involving parallelist constraint based evaluation. The standard DM framework and the OT framework are often accepted as having conflicting assumptions, with Embick (2010:*ix*) explicitly stating that DM 'is incompatible with the dominant view in phonological theory, Optimality Theory (OT), which posits competition among infinite sets of complex objects' (similar sentiments are observed by Tucker 2011:200). Strict versions of DM – especially as formalized by Embick & Marantz (2008) and Embick (2010) – reject parallelism and constraint-based evaluation, as well as blocking effects and optimization. Across morphological theory generally, Embick (2010:11) refers to a **Globalist/Localist Schism**, where Globalist theories of morphophonology such as OT 'require competition between multiple potential expressions of a given input', whereas this is impossible under Localist theories 'because the competitors are not derived by the grammar'. In general, constraints and their 'repairs' play no role in the majority of recent work in DM, largely due to not being pertinent to the specific phenomenon at hand (e.g. Matushansky & Marantz eds. 2013¹; Salzmann 2013; Haugen & Siddiqi 2013; Harley 2014; Gribanova 2015; Watanabe 2015; Moskal 2015; Moskal & Smith 2016; Deal 2016; Kinjo & Oseki 2016; Saab & Lipták 2016; Martinović 2017; a.o.).

2.4. OT-DM

A different set of data and conspiracies has generated the opposite response in other DM practitioners who have sought to incorporate OT into DM architecture. I refer to this model as **Optimality Theoretic Distributed Morphology (OT-DM).** Although there has been quite limited cross-pollination between the OT and DM frameworks, there was early research tying them together, noting in particular the fact that they emerged in the same (generative) linguistic period (e.g. Noyer 1992, 1993, 1994; Bonet 1994; see Wolf 2008:141). For example, some of the earliest integration include Noyer (1992) involving DM-style impoverishment 'achieved by the interaction of inviolable surface constraints and universal feature hierarchies' (Trommer 2001a:71). Trommer's own work (2001a, 2001b, 2002) such as his *Distributed Optimality* model argue that 'crucial mechanisms of DM can be replaced profitably the interaction of violable constraints as in OT' (2001a:12). Models such as Rules & Constraints DM later proposed by Arregi & Nevins (2012) fall under what Trommer calls the *Filter & Hierarchy* approach to DM, a 'filter' being an inviolable (markedness) constraint. Trommer supports his program by citing a number of conspiracies, e.g. a conspiracy involving argument marking of transitive predicates in Dumi [dus] (van Driem 1993;

¹ OT plays a small role role in Matushansky & Marantz (eds.) (2013), namely in Bonet's chapter (Bonet 2013).

Trommer 2001a:66-67,81-83,404-412), as well as cross-linguistic conspiracies such as an anti-homophony constraint resulting in clitic substitution in Italian but deletion in Spanish (Grimshaw 1997, Trommer 2001a:14).²

One recent OT-DM paper involving a morphological conspiracy is presented in Dawson (in press) for Tiwa [lax] (Sino-Tibetan: India). In Tiwa, Dawson identifies that verb roots are prosodically bound, requiring material to their right in the same prosodic word. She formalizes this morphophonological requirement with a markedness constraint *BARE-V. In nearly all contexts, the verb appears with an overt suffix due to independent morphosyntactic properties, and thereby vacuously satisfies the *BARE-V constraint. One exception is with auxiliary verb constructions where an auxiliary selects a bare verb, in violation of *BARE-V. To remedy this, Tiwa employs two repairs in free variation. In one, the verb and the auxiliary integrate and form a single complex phonological word, with the verb cliticizing to the right-adjacent auxiliary. I refer to this as *Verb Cliticization*, shown in (4)a. Dawson argues that the phonological wordhood of the V+Aux complex is supported by the fact that certain word medial consonants in this complex undergo intervocalic voicing. Further, in a second repair, a morphosyntactically higher Focus head which canonically attaches to the right edge of the language attaches instead to the verb. Dawson calls this *Focus Drift*, shown in (4)b. She provides a number of arguments that the position of the focus clitic does not express additional focus semantics on the verb, suggesting post-syntactic reconfiguration.

(4) Tiwa repairs for constraint *BARE-V

```
a. Verb Cliticization: Complex phonological word formation V Aux Foc → (V=Aux=Foc)
(<sub>ω</sub> lí=thái-do=sê)
lí thái-do =sê
go AUX-IPFV =FOC
'he is still going'
b. Focus Drift: Cliticization of focus onto a verb
V Aux Foc → (V=Foc) (Aux)
(<sub>ω</sub> lí=sê) (<sub>ω</sub> thái-do)
```

lí =**sê** thái-do go =**FOC** AUX-IPFV 'he is *still* going'

These repairs form a conspiracy, as in both cases the output is a verb which receives an appropriate 'host' and thereby satisfies *BARE-V. Dawson shows that these data cannot be reduced simply to DM operations Lowering or Local Dislocation (as well as prosodic inversion), and argues that the data necessitates a model that 'explicitly incorporates the constraint-driven component into DM'.

Independently, Foley (to appear) recently discusses a morphological conspiracy involving multiple exponence in Georgian verbal paradigms, part of a larger program *Optimal Vocabulary Insertion*. Foley adopts an OT-DM analysis and shows that although an alternative classic DM account can 'captures the facts, ... it has little explanatory power [as] the collection of impoverishment rules makes no reference to multiple exponence' directly (a parallel line of argument ion is adopted later for Degema).

There are several other works which envision more broadly some type of interaction between DM and OT components, including Haugen (2008, 2011), Wolf (2008), Lahne (2010), Tucker (2011), Sande (2017), and Brown (2017), with Impoverishment in particular being implemented within an OT-DM model (Don & Blom 2006; Opitz 2008; Keine & Müller 2015). Outside of DM, morphological theory has embraced and contributed to constraint based modeling, summarized in Xu (2016), examples including *Cophonology Theory* (Inkelas & Zoll 2007), *Optimal Construction Morphology* (Cabellero & Inkelas 2013), *Stratal OT* (Kiparsky 2015), a.o. Further, a strong view of constraint-based modeling has been adopted by practitioners of OT Syntax, important works including Grimshaw (1997, a.o.), Legendre et al. (2001), Broekhuis & Vogel (2013), Legendre et al. (2016). a.o.

3. Degema: core data and assumptions

3.1. Inflectional clitics

Degema [deg] is a Benue-Congo language of the Niger-Congo phylum spoken in southern Nigeria.³ It is a headinitial language, and maintains a fairly strict SVO word order in which auxiliaries precede the verb and adjuncts

 $^{^{2}}$ Wolf (2015:385) also cites conspiracies as a deciding factor in employing phonologically conditioned suppletive allomorphy via constraints rather than through arbitrary sub-categorization restrictions.

follow the object (including adverbials, CPs, PPs, and ideophones). Degema possesses a series of inflectional proclitics and enclitics which obligatorily adjoin to the verb. Proclitics principally expone subject agreement, though can additionally index tense/aspect and polarity. Agreement proclitics form two sets. Set 1 are used in positive polarity, non-past tense constructions, and all begin with /m/; set 2 appear elsewhere and are vowel initial other than first person singular. Throughout this paper, clitics are marked with PHI-features and as SET1 or SET2. Third person plural is further split into two sets of markers: one which appears with third person plural human referents [+H], and those which appear with third person non-human referents [-H]. Tone and ATR variations involving these clitics are not indicated in Table 2, though are discussed below.

Infl	1^{st}		2 nd		3 rd			
proclitics	Set 1	Set 2	Set 1	Set 2	Set 1		Set 2	2
SG	me	mi	mu	u	mo		0	
PL	ma		ma	0	+H	-H	+H	-H
	me e		ma	a	me	mi	e	i

Table 2: Degema proclitics⁴

Degema has two inflectional enclitics which expone aspect. Factative aspect $\circ = \overline{Vn}$ FAC express perfective aspect / past tense with eventive verbs and present tense with stative verbs. This clitic co-occurs with a high tone on the verb root, and copies the final vowel of the verb if it is vowel final. Additionally, the perfect aspect marker is $\circ = t\overline{e}$ PRF. Kari (2002a; 2002b; 2002c; 2002d; 2003; 2005) presents arguments that both proclitics and enclitics are clitics and not affixes, a position I adopt here but do not elaborate on or defend.⁵ Clitics are illustrated in (5).

(5) Degema inflectional clitics⁶

a.	me =síré		
	1sg.set1=run		
	'I am running' (Kari	2004:286)	
b.	Ohoso o =sá=n		ēnám
	Ohoso 3SG.SET	2=shoot=FAC	animal
	'Ohoso shot an anima	al' (Kari 2004: 2	70)
c.	mi=dé=té	ósama	
	1sg.set2=buy=prf	dress	
	'I have bought a dress	s' (Kari 2004: 29	93)

Cf. *Ohoso Ø sá=n ēnám / *Ohoso o=sá Ø ēnám

Cf. *Ø dé=té ósama / *mi=dé Ø ósama

The only elements which may intervene between the proclitic and the lexical verb are functional auxiliary verbs, a closed class with 12 members. Auxiliaries are outside of the scope here (although see fn X).

Further, the only elements which may intervene between the enclitic and the lexical verb are prosodically light object pronouns. Example (6)a illustrates that when a verb and aspectual enclitic appears with a bisyllabic pronoun

³ Data for this paper comes from the extensive publications on Degema by native speaker-linguist Ethelbert E. Kari (Kari 1997, 2002a, 2002b, 2002c, 2002d, 2003a, 2003b, 2004, 2005a, 2005b, 2006, 2008, 2015), as well as ongoing joint collaboration. Additional consultation with a native Degema speaker was done by the author, summer 2017 in Port Harcourt, Nigeria. Degema has two dialects: Usokun and Atala (also called 'Degema Town'). The current paper is based on the Usokun variety only. Information on the Atala dialect is found in Offah (2000), which reveals a different distribution of clitics (see especially pps. 7,30,33,46-48,57,66-70,79; email me for a copy).

⁴ The distribution of these sets is more complex than this but falls outside of the scope of this paper. What is important to note is that these two sets are in syntactic complementary distribution. Further, in negative-imperative sentences, the second-person singular subject clitic is exponed as e/e=; a marginal subject clitic a= also exists, but only attaches to the bound copular verb bo 'be present' (Kari 2004, 2008: 21); and the non-human variant in 3PL also appears with mass nouns [+M], which have no singular/plural morphological distinction.

⁵ Degema has four additional verb-adjacent enclitics which do not expone aspect, not discussed here. These are =tu 'don't do X', =munu 'stopped X', =vire 'too much', and =ani 'please' (Kari 2004:340).

⁶ Degema orthography is consistent with the IPA, with the following language-specific conventions: $\langle b \rangle = /b/$, $\langle d \rangle = /d/$, $\langle nw \rangle = /\eta^w/$, $\langle ny \rangle = /j/$, $\langle y \rangle = /j/$, $\langle \bar{n} \rangle = /\eta/$, and $\langle v \rangle = /\beta/$. A dot under a vowel indicates retracted tongue root [-ATR], and no dot under the vowel indicates advanced tongue root [+ATR]. ATR is only marked on the first vowel of the word, though ATR harmony applies to all vowels within the word. A high tone is indicated by an acute accent $\langle V \rangle$, a downstepped high is indicated by a macron $\langle \bar{V} \rangle$, and a low tone is not marked.

oyi 3SG 'him/her/it' or *eni* 1PL 'us', the enclitic adjoins to the verb. In contrast, (6)b-c show that if the verb appears with a monosyllabic pronoun, the enclitic must attach to the right edge of that pronoun, and not directly next to the verb. Other orders are ungrammatical.

(6) Surface position of enclitics with monosyllabic object pronoun

a.	Osoabo	o=kótú =n		óyi	Cf. *Osoabo o=	⊧kótú óyi =n
	Osoabo	3SG.SET2=	call=FAC	him/her		
	'Osoabo cal	led him/her	' (Kari 20	04:113)		
b.	o=kótú	wó́ =ō	n		Cf. *o=kótú=n	wō
	3SG.SET2=ca	all you=I	FAC			
	'(s)he called	l you' (Kari	2004: 27	6)		
c.	o=gídí	1	ọáw =tē		Cf. *o=gídí =té	<u></u> bāw
	3SG.SET2=lo	ook.for	them=PRI	Ŧ		
	'(s)he has lo	oked for the	em' (Kari	2004: 282)		

These facts are summarized in the table below. The shading indicates the pattern where the enclitic attaches to pronoun rather than the verb.

	1	2	3	XP {NP/CP/PP/etc.}
SC	méē/mę́ē	wóō	<u>o</u> yí	
SG	V pron=CL	V pron=CL	V=CL pron	V-CL VD
DI	ení	máāny/mą́āny	þáāw/þáāw	V-CL AF
PL	V=CL pron	V pron=CL	V pron=CL	

Table 3: Attachment of enclitic with object pronouns

3.2. Serial verb constructions

Serial verb constructions (SVCs) are a widespread linguistic feature of West Africa (Ameka 2005) and are used extensively in Degema. Non-formally, they are defined as a sequence of more than one verb in a single clause which share verbal arguments and functional categories. In Degema, SVCs are used to express event exhaustion/completion, directionals, benefactives, verbal comparison, comitatives, instrumentals, accompanimentals, refusal, simultaneousness, abilitatives, consequentials, and event coordination (resultatives and purposives are not expressed through SVCs; see Kari 2003a, 2004 for further details).

When inflectional clitics appear in SVCs, two patterns in complementary distribution are seen. One is the **double-marking pattern** in which each verb in series is marked with an identical proclitic and enclitic respectively, in (7)a. A second is the **single-marking pattern** in which only the first verb is marked with a proclitic while only the last verb is marked with an enclitic. This 'bookending' pattern is shown in (7)b.

(7) Clitic patterns

Double-marking SVC patter	ern			
o=sóm=n	úsī	o=túl		wó́ =ōn
3SG.SET2=be.good=FAC	beauty	3 SG.SET	2=reach	you=FAC
'He is as handsome as you.	' (Kari 20	04:157)		
Single-marking SVC patter	'n			
Ohoso o =sóm	túl=	=n	óyi	
Ohoso 3 SG.SET2=be.good	rea	ch=FAC	him	
'Ohoso is as handsome as l	nim.' (Kari	2004:156)	
	Double-marking SVC patter o=sóm=n 3SG.SET2=be.good=FAC 'He is as handsome as you. Single-marking SVC patter Ohoso o=sóm Ohoso 3SG.SET2=be.good 'Ohoso is as handsome as h	Double-marking SVC pattern o=sóm=n úsī 3SG.SET2=be.good=FAC beauty 'He is as handsome as you.' (Kari 200 Single-marking SVC pattern Ohoso o=sóm túl= Ohoso 3SG.SET2=be.good read 'Ohoso is as handsome as him.' (Kari	Double-marking SVC patterno=sóm=núsīo=túl3SG.SET2=be.good=FACbeauty3SG.SET'He is as handsome as you.' (Kari 2004:157)Single-marking SVC patternOhoso o=sómtúl=nOhoso 3SG.SET2=be.goodreach=FAC'Ohoso is as handsome as him.' (Kari 2004:156)	Double-marking SVC pattern $o=s \acute{o}m=n$ $\acute{u}s \vec{i}$ $o=t \acute{u}l$ $3sG.sET2=be.good=FAC$ beauty $3sG.sET2=reach$ 'He is as handsome as you.' (Kari 2004:157)Single-marking SVC patternOhoso $o=s \acute{o}m$ t $\acute{u}l=n$ $\acute{o}yi$ Ohoso $o=s \acute{o}m$ t $\acute{u}l=n$ $\acute{o}yi$ Ohoso $3sG.sET2=be.good$ reach=FAChim'Ohoso is as handsome as him.' (Kari 2004:156)

The double-marking SVC pattern occurs when there is an intervening DP between the two verbs, i.e. the object $us\bar{i}$ 'beauty' in a. In contrast, the single-marking pattern occurs when there is no such intervening object as in b.

The only exception to this generalization involve prosodically light object pronouns. When these light pronouns appear between the verbs, the single-marked SVC pattern occurs despite this intervening element. For example, in (8)a the object pronoun is bisyllabic ϕyi 'her/him' and the SVC displays double-marking. In contrast, in b the object pronoun is monosyllabic *me* 'me' and the SVC displays single-marking (recall from Table 2 that these same light pronouns also have a different distribution with respect to enclitics).

(8)

Double-marking pattern with bisyllabic pronoun a. mi=dúw=n óvi mi=tá=ān 1SG.SET2=go=FAC **1SG.SET2**=follow=FAC her/him 'I went with her/him' (Kari 2004: 201) Single-marking pattern with monosyllabic pronoun b. Breno o=dúw mé tá=**ān** Breno **3**SG.SET2=follow me go=FAC 'Breno went with me' (Kari 2004: 115)

In the double-marking environment where there is an intervening object (other than a prosodically light object), the single-marking pattern is ungrammatical here and both sets of clitics are obligatory.

(9) Ungrammatical single-marking in presence of intervening object

a.	Tatane o=kótú=*(té)	éni	*(o)=kpérí=tē	ínúm
	Tatane 3SG.SET2=call=*(PRF)	us	*(3SG.SET2)=tell=PRF	something
	'Tatane has called us and told (us	s) som	nething' (Kari 2003a: 285)	
b.	Tatane o=sá=*(n)	ēnán	n *(o) =gbíyé=ēn	
	Tatane 3SG.SET2=shoot=*(FAC)	anim	hal *(3SG.SET2)=kill=FA	AC
	'Tatane shot an animal dead' 'Ta	atane	shot and killed an animal'	(AUTHOR $201X \cdot XX$)

In contrast, in the single-marking environments where the verbs in series appear without an intervening object or only with a light object, the double-marking pattern is ungrammatical for some speakers but questionable for others. This variation is detailed in AUTHOR (201X:XX). The value ^(?) is acceptable but dispreferred, [?] is unnatural and dispreferred, and ^{?*} is grammatically questionable; the slash indicates two different speakers intuitions. Acceptability increases with a prosodically light pronoun (b.). In general, examples with the factative enclitic =Vn are more accepted than with perfect $=t\bar{e}$.

(10) Double-marking pattern interpretations (AUTHOR 201X:X)

a. ^{?/?}*Breno o=síré=ēn o=tá=ān Breno 3SG.SET2=run=FAC 3SG.SET2=go=FAC 'Breno ran and went'
b. ^{(?)/?}Breno o=vón mé=ēn o=yí=īn Breno 3SG.SET2=take me=FAC 3SG.SET2=come=FAC 'Breno brought me'

The interpretations of these examples contrasts with those in (9) where single-marking is strongly rejected as ungrammatical. Even for those speakers who may more readily accept a double-marking pattern here, it is the more marked variant of the two, reflected by its virtually non-existing status in the existing documentation work.

Moreover, further consultation with a native speaker shows that he unequivocally rejects double-marking in single-marking context (11).

(11) Ungrammaticality of double-marking in the single-marking context

a.	Ohoso	₀=tá(*= n)	. (* o=)dé=n	isen
	Ohoso	3SG.SET2=go(*=FAC) (*3sg.set2=)buy=FAC	fish
	'Ohoso v	vent and bought fi	sh' [ohk_201'	707xx]	
b.	Breno o	=dúw	mé(*= n)	(* • • =)tá=ān	
	Breno 3s	SG.SET2=follow	me(*=FAC)	(*3SG.SET2=)go=FA	AC
	'Breno w	vent with me' [ohk	c 201707xx]		

Furthermore, for all speakers it is ungrammatical to have only one medial clitic in the single-marking context.

(12) Ungrammaticality of marking only one verb with double-marking

a. Ohoso o=sóm (*o=)túl=n óyi Ohoso 3SG.SET2=be.good (*3SG.SET2=)reach=FAC him/her 'Ohoso is as handsome as him.' (E.E. Kari p.c., 20XX XX XX)

b.	Ohoso	o=yí(*=té)	kótú=té	óyi
	Ohoso	3SG.SET2=come(*=PRF)	call=PRF	him/her
	'Ohoso has	s come and called him' (Kan	ri 2003a: 28	85)

3.3. Syntax of SVCs

I adopt a modified version of Collins' (1997, 2002) analysis of SVCs involving nested vP shells where V₁ selects v_2P as its complement: [ASPP ASP° [$_{vP}v_1^\circ$ [V₁° [$_{vP}v_2^\circ$ [V₂°]]]]. This vP complementation structure is illustrated in Figure 1.



Figure 1: vP complementation structure: Asp $> v_1 > V_1 > v_2 > V_2$

In this structure, only one set of functional heads occur, i.e. a single Asp^o head. The external argument (the subject) is base-generated in spec v_1P with a corresponding empty category e in spec v_2P (the latter's technical identity is outside the scope of this paper). Internal arguments (objects) are base-generated in positions within VPs.

Following Collins, in order to license multiple verbs under a single set of verbal projections which are not within coordinate or subordinate structures, Collins posits a *Serialization Parameter*, which states that a functional head I°/T° can license multiple Vs (Collins 1997: 493), and later modifies this as a little v° able to license multiple Vs (Collins 2002: 8). In the version adopted here, I assume that the functional head Asp° (or the functional field as a whole) license multiple vPs. A similar structure is assumed in Carstens' (2002) analysis of Ijo SVCs.⁷ Further, following much work in DM, I assume that the Agr features which are exponed as the subject agreement proclitics are inserted post-syntactically (Embick & Noyer 2007, Kramer 2010, Norris 2014, a.o.). Therefore, no agreement (Agr°) projections appear in the syntax which would correspond to the subject agreement proclitics.

I adopt that both the single-marking and the double-marking SVC clitic patterns have the syntactic structure in Figure 1, and argue that the differences between these patterns emerge only post-syntactically. There is no positive evidence that these different surface patterns involve distinct numeration types or distinct sets of syntactic operations. Both patterns involve the same types of functional categories, and both can be used with any type of

⁷ Several alternative proposals for SVC syntax exist. One involves double-headedness and parallel structure (Baker 1989, Hiraiwa & Bodomo 2008), in which an object is simultaneously an argument of both V_1 and V_2 . Another involves VP adjunction where v_2P/V_2P is adjoined to the first verb projection (Baker & Stewart 2002). A third involves Aboh's (2009) proposal that in SVCs, V_1 is a functional projection while V_2 is a lexical projection. As the stated goal of this paper is to derive the distribution of the clitic patterns in SVCs post-syntactically, I do not present further arguments against these alternative structures, and maintain that whichever syntactic structure is chosen, it will be inadequate for accounting for the distribution of clitics in full.

lexical verb regardless of transitivity or other syntactic-semantic distinctions. To illustrate, the SVC duw ta 'go with' (lit. follow go) involves a transitive and intransitive verb. If the object of the transitive verb duw 'follow' appears insitu between the verbs, the double-marking pattern occurs as in (13)a. In contrast, if the object is moved e.g. under \bar{A} -movement, the verbs become surface adjacent and the single-marking pattern occurs instead, shown in b.

(13) Syntax-equivalency of double-marking and single-marking (AUTHOR XX:XX)

a.	mi=dı	úw=n	óvo	o r	ni≓tá=an	?
	1SG.S	ET2=f	ollow=FAC wh	0 1	SG.SET2	=go=FAC
	'I wei	nt wit	n who?'			
b.	ovó _i	nų́	mi=dúw	e	vó i	tá=an?
	who	that	1SG.SET2=foll	low v	vho	go=FAC
	'who	did I	go with?'			•

Further, the number of verbs in the SVC also play no role in determining these patterns, as shown in (14) involving five verbs in series in a single-marking pattern.

(14) Ohoso o=tá dé vó yí kíyé=n óyi
Ohoso 3sG.SET2=go buy take come give=FAC him/her
'Ohoso went and bought (something) and brought (it) to him/her' (Kari 2004: 121)

I return to these facts below in Section 6.1 Alternative 1 – Syntactic verb movement.

Finally, an important note needs to be made regarding aspect in SVCs. In the examples provided throughout, in the double-marking pattern the verbs share identical clitics, e.g. in (13)a above. However, in exactly one context the verbs bear different clitics. Recall that proclitics primarily expone subject agreement but the distribution of set 1 versus set 2 is conditioned by tense/aspect/polarity. In SVCs which express imperfective aspect (habitual or present continuous), the first verb is marked with a set 2 proclitic and the second verb is marked with a set 1 proclitic. This non-identical double-marking is found regardless of the presence of an intervening object.

(15) Obligatory non-identical proclitics in double-marking pattern expressing imperfective aspect

a.	tẹvúr	ó tevuro	o =rékéréké	n	nō=dí	á	-	-
	every	/day	3SG.SET2=be	slow 3	SG.SET1=eat	NPM		
	'ever	yday, she e	ats (them) slow	ly' (<mark>AUT</mark>	HOR 201X:X	XX)		
b.	eni	e=dúw=r	1	óyi	mę́ =tá			
	we	1pl.set2	efollow=FAC	him/he	er 1PL.SET1	=go		
	'we a	are going w	ith him' (AUTH	IOR 2012	X:XX)			

- c. Osoabo o=kótú=n óyi mó=kpérí ínúm
 Osoabo 3SG.SET2=call=FAC him 3SG.SET1=tell something
 'Osoabo called (him and is) telling him something.' (Kari 2004: 113)
- d. Breno **q**=dí=n ídíyōm **q**=vó **mó**=réré mú úsuweny Breno **3**SG.SET**2**=eat=FAC food **3**SG.SET**2**=take **3**SG.SET**1**=walk on road 'Breno is eating while working', 'Breno is eating and walking on the road' (Kari 2003a: 282)

In contrast, SVCs expressing perfective and perfect aspect or negative polarity use identical set 2 proclitics, while SVCs expressing future tense use identical set 1 proclitics, shown in (16). In mono-verbal clauses, both imperfective aspect and future tense are marked solely with a set 1 proclitic resulting in surface ambiguity between these tense/aspect distinctions.

(16) Future tense single- and double-marking with set 1 proclitics (AUTHOR 201X:XX)

a.	Bren	o mó =ḍúw	mé	Ø	tậ	
	Bren	o 3sg.set1=follow	me	Ø	go	
	'Brei	no will go with me'				
b.	eni	mé =ḍúw	óyi	mę́⁼	=tá	
	we	1PL.SET1=follow	him	1pl	.set1=go)
	'we v	will go with him/her'				

Given these facts about non-identical proclitics largely conditioned by distinct tense/aspect environments, it would be reasonable to assume that each verb in the SVC has a distinct Asp° projection, which can differentially condition the distinct proclitics.

(17) Asp-Serialization alternative with two Asp heads $\begin{bmatrix} Asp - 1 P \dots \begin{bmatrix} y-1 P \dots \begin{bmatrix} V-1 P \dots \begin{bmatrix} Asp - 2 P \dots \begin{bmatrix} y-2 P \dots \begin{bmatrix} V-2 P \dots \end{bmatrix} \end{bmatrix} \end{bmatrix} \dots$

Despite this plausibility, I do not adopt this position. If there were two distinct Asp projections, we predict that the micro-events expressed by each verb could bear distinct aspectual values, e.g. 'he ate (then) and is coming (now)'. There is little if any evidence for this. For example, in (15) the [set 2=...set 1=...] structure together express imperfective aspect idiomatically, and cannot be decomposed into two independent aspect values whose sum expresses imperfective types habitual or present continuous. For those examples which do exist which are translated with multiple aspectual value (e.g. (15)c), they can easily be interpreted as a single imperfective aspectual value. Further, no instances of mixed enclitics occur (i.e. V_1 with $=V\overline{n}$ FAC and V_2 with =te PRF), which are the primarily exponents of aspect. I assume that when the syntactic features expressing imperfective aspect appear in a SVC, these features are split across the verbs and result in the distinct exponents on the two verbs (set 2 and set 1 respectively). The exact mechanisms which derive this structure fall outside of the scope of this paper.

3.4. Summary of distribution

A summary of the complementary distribution of single-marking and double-marking is provided in Table 4.

V ₁	V_2	Doubl	e-marking	Si	ngle-marking
Ø		?*~*	$cl=[V_1]=cl cl=[V_2]=cl$		$cl=[V_1 V_2]=cl$
Pron _σ		?~*	$cl=[V_1 D_{\sigma}]=cl cl=[V_2]=cl$		$cl=[V_1 D_{\sigma} V_2]=cl$
Pron _{oo}			$cl=[V_1]=cl D_{\sigma\sigma} cl=[V_2]=cl$	*	$cl=[V_1 D_{\sigma\sigma} V_2]=cl$
DP			$cl=[V_1]=cl \mathbf{DP} cl=[V_2]=cl$	*	$cl=[V_1 \mathbf{DP} V_2]=cl$

Table 4: Complementary distribution of clitic marking in SVCs

4. DM analysis: A first pass

The distribution of clitics in SVCs can be summarized as a list of licit and illicit structures, shown in Table 5.

Li	cit structures	Illicit structures		
	cl=[V]=cl	* cl=[V] Ø	* Ø [V]=cl	* Ø[V]Ø
	$cl=[V D_{\sigma}]=cl$	* $cl=[V D_{\sigma}] \mathbf{\emptyset}$	* $\mathbf{\emptyset} [V D_{\sigma}] = cl$	* $\boldsymbol{\emptyset} [V D_{\sigma}] \boldsymbol{\emptyset}$
	$cl=[V_1 V_2]=cl$	* $cl=[V_1 V_2] \mathbf{\emptyset}$	* $Ø [V_1 V_2] = cl$	* $cl=[V_1]=cl \boldsymbol{\emptyset} [V_2] \boldsymbol{\emptyset}$
	$cl=[V_1 D_{\sigma} V_2]=cl$	* $cl=[V_1 D_{\sigma} V_2] \mathbf{\emptyset}$	* $\mathbf{Ø} [V_1 D_{\sigma} V_2] = cl$	* $cl=[V_1 D_{\sigma}] \boldsymbol{\emptyset} cl=[V_2]=cl$
	$cl=[V_1]=cl D_{\sigma\sigma} cl=[V_2]=cl$	* $cl=[V_1] \mathcal{O} D_{\sigma\sigma}$	* $cl=[V_1]=cl D_{\sigma\sigma}$	* etc.
		cl=[V ₂]=cl	$Ø[V_2]=cl$	

Table 5: Licit and illicit SVC clitic patterns

The central generalization is that all verbs in the licit structures appear in a verbal constituent which is marked by a proclitic and an enclitic (note that enclitics are present only when the morphosyntactic features which they expone are present, i.e. factative and perfect but not future). In comparison, all verbs in the illicit structures appear in a constituent which is not fully marked by these inflectional clitics.

We understand these data as demonstrating a **morphological conspiracy** in the sense that all surface structures exhibit verbs or verb complexes which are marked with a full set of clitics, i.e. a structure which converges on a structure CL=[V]=CL. We can understand this structure as adhering to a **morphological well-formedness condition** which can be formalized as a markedness constraint V=WF(INFL) which states that verbs must appear with appropriate inflection.

The constraint V=WF(INFL) results in a number of independent post-syntactic operations, hence the conspiracy. First, it triggers insertion of subject agreement AGR nodes not present in the syntax, exponed as the proclitics. Second, the aspectual enclitics exponing the functional Asp^o head is moved to the verb where it right-attaches. Third, if there is an intervening prosodically heavy object the aspectual enclitic is both lowered onto V₁ and copies onto V₂. Fourth, the fact that adjacent verbs or verbs whose intervening object is prosodically light appear with single marking suggests that they form a constituent at some level of representation. If we accept that these verbs do *not* form a constituent in the syntax (i.e. no v_1 -to- v_2 syntactic head movement), then constituent

formation must take place via a post-syntactic operation, i.e. $(V_1) (V_2) > (V_1+V_2)$. Informally speaking, if the verbs are sufficiently local defined prosodically, they may form a constituent to 'share' the clitic marking resulting in single-marking. If however they are not sufficiently local, each verb receives its own set resulting in double-marking. This is summarized below.

(18) Morphological conspiracy triggered by V=WF(INFL)

- a. Insertion of subject agreement AGR nodes
- b. Lowering of aspectual enclitic to V_1
- c. Copying of aspectual enclitic to V_2
- d. Constituency formation of sufficiently local verbs

This conspiracy is characteristic of morphological well-formedness conditions as discussed in the literature, e.g. Halle & Marantz' (1993: 137) account of English tense morphology requiring a verb for well-formedness.

By definition, DM operations are due to morphological requirements, though they are often not framed as triggered by formal markedness constraints. Two such operations are **dissociated node insertion (DNI)** and **local dislocation (LD)**, which can be used to model the Degema morphological processes sketched above.

4.1. Dissociated Node Insertion (DNI)

Embick & Noyer (2007: 305-310) discuss the insertion of dissociated material at spell-out, which they understand as two types: dissociated feature insertion and dissociated node insertion (i.e. bundles of features on a separate morphological node). For the purposes of this paper, I treat them together and refer to them as dissociated node insertion (DNI). DNI is used to capture what Embick & Noyer call **ornamental morphology**, exemplified with Latin thematic vowels and agreement morphemes (glossed TH and AGR, boldened). Such ornamental morphology is inserted post-syntactically and therefore does not expone syntactic terminal nodes.

(19) Latin insertion of ornamental morphology

laud-**ā**-bā-**mus** root-**TH**-TNS-**AGR** 'We were praising'

DNI of agreement morphology relies on a concept of **feature copying** where a feature 'present on a node X in the narrow syntax is copied onto another node Y at PF' (Embick & Noyer 2007: 309). The exact mechanisms which underlie the agreement patterns are outside of the scope of this paper however (e.g. *Agree-Link* vs. *Agree-Copy* – Arregi & Nevins 2012:86). Post-syntactic insertion of morphological nodes has been well-supported in the DM/Minimalist literature (Embick & Noyer 2007, Kramer 2010, Norris 2014, a.o.).

With respect to the Degema data, subject agreement proclitics and the second aspectual enclitic on V2 are inserted via DNI, resulting in structures which comply with the V=WF(INFL) constraint. These proclitics appearing in both single- and double-marking are the exponents of feature bundles which are copies of subject PHI-features. Example (20) below repeats ex. (9)a above, involving third person singular proper name subject *Tatane* and the verbs *kotu* 'call' and *kperi* 'tell'. DNI also accounts for the presence of a second aspect marker in the double-marking pattern. In this pattern, the first aspectual enclitic on V1 is the exponent of the syntactic Asp° head while the second enclitic on V2 is a dissociated aspectual agreement node. Both of these share the same features and consequently are both exponed by the same vocabulary items. In the example, agreement nodes inserted via DNI are boldened.

(20) DNI of subject agreement in Degema

	0	•		
Input: [AspP	Subject _i [$Asp^{\circ}[_{v1P}]$	Subject _i [$v_1^{\circ}+V_1^{\circ}$ [$[v_2^{\circ}+V_2^{\circ}\dots$
	Tatane	√PRF	$v_1^{\circ} + \sqrt{KOTU}$	$v_2^{\circ} + \sqrt{KPERI}$
Output:	/Subject/	/PRF/	/AGR _{sBJ} /+/Verb ₁ /	/AGR _{sbJ} /+/Verb ₂ /+/AGR _{ASP} /
	/Tatane/	/=te/	/ o= /+/kotu/	/ o= /+/kperi/+/= te /
	Input: [_{AspP} Output:	Input: [AspP Subject _i [Tatane Output: /Subject/ /Tatane/	Input: $\begin{bmatrix} Asp^P & Subject_i & [Asp^o & [_{\nu IP} \\ Tatane & \sqrt{PRF} \\ & & \\ Output: & /Subject/ & /PRF/ \\ /Tatane/ & /=te/ \end{bmatrix}$	Input: $\begin{bmatrix} Asp^{\circ} & Subject_i & Asp^{\circ} & [v_{1P} & Subject_i & v_1^{\circ}+V_1^{\circ} & \dots & \\ Tatane & \sqrt{PRF} & v_1^{\circ}+\sqrt{KOTU} \\ & & & & & \\ Output: & /Subject/ & /PRF/ & /AGR_{sBJ}/+/Verb_1/ \\ /Tatane/ & /=te/ & /o=/+/kotu/ \end{bmatrix}$

This aspectual concord is directly parallel with Kramer's (2010) analysis of Amharic definiteness marking. In (21) below, Kramer argues that the first instance of the definite marker -u is the exponent of D° (and appears right-adjacent via local dislocation), whereas the second -u is the exponent of a post-syntactic Agr node insertion valued through copy featuring (this latter exponent is optional).

(21) Amharic definiteness concord – Dissociated Node Insertion of DEF node tillik'-u t'ik'ur(-u) bet
big-DEF black(-DEF) house
'the big black house' (Kramer 2010: 229)

In both the Amharaic and Degema cases, surface identical morphemes correspond to features which originate in distinct modules in the derivation.

4.2. Local Dislocation (LD)

The DM operation local dislocation (LD) takes two linearly adjacent constituents $(\alpha)^*(\beta)$ and converts them into a single constituent with or without linear changes, i.e. $(\alpha+\beta) \sim (\beta+\alpha)$ (Embick & Noyer 2001: 561). After LD, morphemes which would otherwise be expected to be independent words on syntactic criteria (i.e. expone syntactic terminal heads) are converted into having an affixal relationship. Several accounts employ LD to account for the position of clitics, e.g. in (22) the Latin conjunct clitic =que local-dislocates and attaches to the right edge of the first word of the second conjunct.

(22) Local dislocation with Latin =que

 $(que)^*(barbaris)... \rightarrow ...(barbaris+que)...$ [barbaris = que equitibus paucis]barbarian.ABL.PL=and cavalry.ABL.PL few.ABL.PL '...and a few barbarian cavalry-men' (Embick 2007b)

LD operates on linear order and is therefore distinct from the DM operation *Lowering* which is sensitive to syntactic hierarchical structure rather than linear order.

LD can account for several effects of the Degema clitic patterns, illustrated in (23). First, in those structures which appear with aspectual enclitics, these enclitics local-dislocate and right-attach to the adjacent verb; as shown above the enclitics appear in a higher syntactic Asp° position.⁸ Second, as shown in ex. (6) and throughout, prosodically light object pronouns are the only morphemes which intervene between the verb and the enclitic. Here, the light pronoun local-dislocates with the preceding verb, though without any linear order changes. LD takes place here due to the pronoun being prosodically deficient and unable to form a relevant constituent on its own. Third, if two verbs appear surface adjacent they undergo LD and form a single constituent, i.e. a morphological verb compound. Because LD is sensitive to linear rather than hierarchical structure, this captures the fact that any sequence of adjacent verbs undergoes LD even those in which a syntactic object has been \bar{A} -moved or is covert (as shown in (13)-(14) above involving *wh* \bar{A} -movement).

(23) Local dislocation in Degema

a.	(/asp/)*(/V/)	\rightarrow	(/V+asp/)	(/= n /)*(/som/)	\rightarrow	(/som+ n /)	[Ex. (7)a]
b.	$(/V/)*(/D_{\sigma}/)$	\rightarrow	$(/V+D_{\sigma}/)$	(/tul/)*(/wo/)	\rightarrow	(/tul+ wo /)	[Ex. (7)a]
c.	$(/V_1/)*(/V_2/)$	\rightarrow	$(/V_1+V_2/)$	(/som/)*(/ tul /)	\rightarrow	(/som+ tul /)	[Ex. (7)b]

If more than one morpheme appears which is applicable, LD takes place iteratively, shown below.

(24) Iterative local dislocation in Degema

a.	$(/asp/)*(/V/)*(/D_{\sigma}/)$	ĺ →	$(/V/+/D_{\sigma}/+/asp/)$	
	(/=te/)*(/gidi/)*(/baw/)	\rightarrow	(/gidi/+/baw/+/te/)	[Ex. (6)c]
b.	$(/asp/)*(/V_1/)*(/D_{\sigma}/)*(/V_2/)$	\rightarrow	$(/V_1/+/D_{\sigma}/+/V_2/+/asp/)$	
	(/=n/)*(/duw/)*(/me/)*(/ta/)	\rightarrow	(/duw/+/me/+/ta/+/n/)	[Ex. (8)b]

LD is therefore used to account for the fact that certain verb clusters and verb+pronoun clusters have the same distribution as single verbs with respect to inflectional morphology, as they each form a morphological constituent. Thus, Degema exhibits two types of local dislocation: prosodically driven LD for $(V)^*(D_{\sigma})$ sequences, and morphologically driven LD for $(V1)^*(V2)$ sequences. If a verb can locally dislocate to satisfy this condition, it does

⁸ Verb movement to Asp° is ruled out because this would predict that in SVCs the aspectual enclitic should always appear between V1 and V2, contrary to fact.

so, avoiding the need for unnecessary dissociated node insertion. In this way, the double-marking pattern represents a 'repair' when a single-marking pattern is not possible.⁹

4.3. Support for (V₁+V₂) morphological constituency: Evidence from tone

Support for the morphological constituency of verb clusters comes from grammatical tone. Degema contrasts /H/ vs. /L/ tone, as well as downstepped /H/s (downstepped Hs are transcribed as [*H] and in orthography with a macron e.g. $\langle \bar{a} \rangle$). Tone on verbs is predictable from grammatical context. In both the factative and the perfect aspect, a grammatical H tone is assigned to the verb, book-ended by a set 2 agreement proclitic with low tone and the aspectual enclitic $= \bar{Vn}$ or $= t\bar{e}$, as in (25)a. The final downstepped H associated with the enclitic only surfaces if it is followed by all high tones b., or is at the end of a phonological phrase (c.).

(25) Grammatical all High tone between clitics

a.	mị=(ḍé)=n	ágada			
	1SG.SET2=	ouy=FAC chair			
	'I bought a	chair' (E.E. Kari, p	.c. XXX	.)	
b.	Tatane	o=(kótú)=té	éni	ọ=(kpérí)=tē	ínúm
	Tatane	3SG.SET2=call=PR	F us	3SG.SET2=tell=PRF	something
	'Tatane has	s called us and told ((us) som	ething' (Kari 2003a:	285)
c.	o=(kótú	w ∕į)=ōn			
	3SG.SET2=c	call you=FAC			
	'(s)he calle	d you' (Kari 2004: 2	276)		

Crucially, in the single-marking SVC all verbs and prosodically light object pronouns between the clitics are marked with H tone as well, as in (26). Unlike verbs, objects such as /isen/ [isèn] ([isén] in isolation) in b. and subjects such as Ohoso /óhósò/ in c. fall outside of this grammatical H tone.

(26) Grammatical all High tone marking between clitics in a SVC

a.	Breno o=(d	lúw	mé	tá)=	ān			
	Breno 3SG.	set2=follow	me	go=I	FAC			
	'Breno wen	t with me' (Kari	2004	: 115)			
b.	ọ=(tá	₫é)=n	isen					
	3SG.SET2=g	buy= FAC	fish					
	'(s)he went	and bought fish'	(Kar	i 200	4: 311)			
c.	Ohoso	ọ=(tá	dé	vộ	yį	kíyé) =n	óyi	
	Ohoso	3SG.SET2=go	buy	take	come	give=FAC	her/him	
	'Ohoso wen	t and bought (so	meth	ing) a	nd brought	(it) to her/	'her' (Kari 2004: 1	21)

The facts with future tense are similar. This tense/aspect is marked via a set 1 proclitic but no enclitic, and all verbs are marked with grammatical H tone. In the factative and perfect in (25)-(26) above the set 2 proclitic was marked

(i) Verb cluster formation in Swiss German ...**laat**_{V1}...[la_{V2} ...[gaa_{V3} ...] \rightarrow ...[[$la_{V2}=gaa_{V3}$]+**laat**_{V1}] das er si la_2 gaa_3 **laat**₁ that he her let.PARTICLE go.INF **let.3SG** 'that he lets her go' (Salzmann 2013: 89)

Salzmann argues that "verb cluster formation + inversion is a late PF-process akin to Local Dislocation" (p. 90), and that to account the cluster formation via syntactic head-movement would result in a number of problems such as not making the correct predictions regarding cluster impenetrability facts (p. 114).

⁹ In the DM literature, LD most commonly occurs with functional material and not two lexical morphemes such as verbs. However, there is precedence for post-syntactic V+V cluster formation from Germanic. Salzmann (2013) demonstrates that with Swiss German verbs which appear with verb phrase complements, the verb phrase complement is by default linearized to the right, unlike DPs which are linearized to the left (p. 91). However, at the surface level, the selecting verb which is morphosyntactically highest appears at the right edge of a verb cluster, the so called $123 \rightarrow 231$ verb cluster change, shown below:

with L tone. In the future tense/aspect below, the set 1 proclitic is marked with H tone with the exception that the proclitic me= 1sg.set1 is always marked with /L/ tone.

(27) Grammatical /H/ tone with future tense/aspect

a.	me=(yí)	<u></u> bána	úkpé	vivi			
	1SG.SET1=come	here	year	another			
	'I will come here	another ye	ear/next yea	ar' (Kari 20	04: 230)		
b.	mó=(vón)	íva báav	v mó=	(vó	yókúró)		
	3SG.SET1=take	two then	n 3sg.	SET1=take	leave		
	'(s)he will leave	with both c	of them', '(s)he will ta	ke two of them	and leave' of	hk 20170806]

Example b. demonstrates that all verbs within the SVC are marked with H tone.

We can contrast the behavior of verbs in SVCs with that of transitive verbs with overt object nouns. Nouns have lexical contrast as well as being subject to grammatical tone changes. Nearly half of all Degema nouns surface with a [LⁿH] tone pattern in isolation (~45%), which is the default tone pattern in the language also found on verbs in isolation. Given that this pattern can be predicted from default rules, I analyze such nouns as inherently toneless, i.e. $/\emptyset$. Other nouns show a restricted number of lexical patterns, all of which must contain at least one lexical /H/ tone. Data in (28)a-b demonstrate that in a SVC where the last verb is three syllables long e.g. *yokuro* 'leave', the /H/ tone spreads across it to the end of the domain, resulting in a consistent all H pattern across the SVC. In contrast, the data in (c.) show that when the SVC is followed by a three syllable toneless *noun* /agada/ 'chair', the /H/ tone does not spread to the end. The noun surfaces as [ágàdà], with a /H/ assigned to the initial syllable of the object.¹⁰

(28) Tonal differences in future tense/aspect: Verbs vs. objects

a.	Tense/aspect H tone	e spread acro	ss toneless cor	stituent $(V_1+V_2+V_3)$		
	mộ=(vón vớ	ģ yộkúró)		[jókóró]	(cf. *yokuro	*[jókừrò])
	3SG.SET1=take ta	ke leave				
	'(s)he will take (it)	and leave', '(s)he will leave	e with (it)' [ohk_20170)806]	
b.	Tense/aspect H tone	e spread acro	ss toneless con	stituent ($V_1+O_{\sigma}+V_2$)		
	Bréno mó=(dúw	mé	yókúró)	[jókúró]	(cf. *yokuro	*[jókùrò])
	Breno 3SG.SET1=fo	ollow me	leave			
	'Breno will leave w	ith me' [ohk	20170806]			
c.	Tense/aspect H tone	e spread acro	ss toneless cor	stituent (V_1+V_2) to ex	clusion of followi	ng toneless (O)
	Bréno mó=(tá	gén)	(ágada)	[ágàdà]	(cf. *ágádá	*[ágádá])
	Breno 3SG.SET1=ge	o look.at	chair			
	'Breno will look at	a chair' <mark>[ohk</mark>	20170806]			
			- 1			

Finally, negation is expressed via a /H/ tone on a type 2 proclitic and an all /L/ pattern on the verb. No aspectual enclitics occur and the distinction between perfective, perfect, imperfective, and future is collapsed.

'(s)he will go look at an animal' [ohk_20170806]

¹⁰ There is a complication with the assignment of grammatical tone: when the last verb appears with a two syllable toneless object and this object appears last in the phonological phrase, the second verb is downstepped. This is unexpected given that no downstepped H is present when an object is not present (ii), and because the downstepped H appears between the verbs. [The superscript ^x here indicates not attested.]

⁽ii) mộ=(**tá gẹn**) ệnám [mó tá ⁺gén énám] (Cf. ^x[mó tá gén énám] ^x[mó tá gén ⁺énám] 3SG.SET1=**go look.at** animal ^x[mó ⁺tá gén énám])

As mentioned above, downstepped Hs only surfaces if they appear following all high tones or they are at the end of a phonological phrase.

A partial answer to this unexpected data involves restrictions in the tonal grammar of Degema. Of a corpus of Degema nouns from Kari's (2008) dictionary, there are downstepped surface patterns $[H^+H]$, $[H^+HH]$, $[H^+HH]$, $[LH^+H]$, $[LH^+H]$, $[LH^+H]$, $[LLH^+H]$, but there are no patterns ${}^{x}[(L)HH^+HH]$ where the word has two H tones followed by a downstepped H-H sequence. I therefore suspect that phonological markedness constraints affect the distribution of grammatical H tone with future tense/aspect.

(29) Grammatical /L/ tone expressing negation

a. mée mi=tá=an do mée mí=(mon) óyi [...mí mòn ójì] I 1SG.SET2=go=FAC but I NEG\1SG.SET2=see her/him 'I went but I didn't see her/him' (Kari 2004: 138)
b. mí=(seneke) mú ívom NEG\1SG.SET2=think in inside 'I don't think so' (Kari 2004: 32)

As with grammatical tone above, the /L/ tone spreads across the entire SVC constituent.

(30)

a.	osamá yo	ó़=(buw		kel		me)	[ś bùw	kèl mè]
	shirt the	NEG\3SG.S	ET2=be.b	ig be.	more.than	me		
	'the shirt is	not bigger	than me' [i.e. not ov	ersized] (Ka	ari 2004: 1	55)	
b.	ó=(deri	me)	kậbuló	ó=(mem	e di)	įdíyóm	yọ	
	NEG\3SG.SE	ET2=know	me beca	ause NEC	3SG=agre	e eat	food the	
	'(s)he refus	sed to eat the	e food bec	ause (s)he	doesn't kno	ow me' (Ka	ari 2004: 45	5)
c.	Ohoso ó=(l	kotu	me	kake)	inum			
	Ohoso NEG	3sg.set2=	call me	show	something	g		
	'Ohoso did	not call me	and (did	not) show	(me) somet	hing' (Kari	i 2003a: 278	3)

Parallel to the examples in (28), if the negative SVC is followed by a three-syllable toneless object, for some speakers the noun bears an initial /H/ tone (a.). For other speakers, inherently toneless objects remain toneless under negation regardless of syllable count and bear all /L/ tones on the surface b. (the [%] indicates inter-speaker variability).

(31)

a.	[%] Breno	ó=(meme	mene)	úmene	
	Breno	NEG\3SG.SET2=	agree do=	work	
	'Breno ref	fused to work' (K	ari 2004: 117)		
b.	[%] Breno	ó=(meme	mene)	umene	[ohk_20170806]

In contrast, when the verbs in the SVC are separated by a prosodically heavy object resulting in double-marking, only V1 bears negation tone and does not spread across the intervening object, shown below. Any subsequent verbs are marked with /H/ tone and a set 2 proclitic bearing /L/ tone (no enclitics occur here, as well). Although negation tone is marked on only V1, negation takes scope over both verbs (Kari 2004: 111).

(32)

a.	Ohoso	ó=(kotu)	éni g	=(káké)	ínúm
	Ohoso	NEG\3SG.SET2=	call us 3	SG.SET2=show	something
	'Ohoso di	d not call us and sl	now (us) son	nething' (Kari 20	03a: 279)
b.	Osoabo	ó=(von)	élege	ọ=(fíyá)	
	Osoabo	NEG\3SG.SET2=	take knife	3sg.set2=cu	ıt
	'Osoabo d	lid not use a knife	o cut somet	hing' (Kari 2004:	111)
	(Not 'Osoa	abo cut something	but didn't u	se a knife (to do i	it)')
c.	Cf. Ungra	ammatical 'doublir	g' of negati	on tone	
	*Osoabo	ó∕=(von)	élege	ó≠(fiya)	

This example illustrates that V2 can only bear grammatical tone expressing negation only when it is sufficiently local to V1, which supports cross-module constituency of these verbal sequences.¹¹

¹¹ Another set of complications for grammatical tone and morphological constituency comes from Degema auxiliaries. Auxiliary verbs are a closed class with 12 members which appear between the subject and lexical verb, consisting of: *nwanyki* 'intended to', *manyki(ma)* 'shouldn't', *ma₁* 'would have', *ku* 'did' (verum focus), *buka(ma)* 'begin', *ma₂* 'not yet', *da₁* 'then', *si₁* 'had finished', *kiri/kuru* 'again, also', *da₂* 'about to', *si₂/su* 'still', and *ga* 'actually'. Generally speaking, auxiliary constructions show properties which can be subsumed under serial verb

4.4. Problems with Rules & Constraint DM

The DM operations Dissociated Node Insertion (DNI) and Local Dislocation (LD) have been used to account for morphological adjustments necessary to comply with the morphological markedness constraint V=WF(INFL). In sections 2.3-2.4, we contrasted two sub-theories Rules & Constraints DM (R&C DM) and Optimality Theoretic DM (OT-DM) which treat DM operations in fundamentally different ways. In this section, I discuss an analysis of the Degema data within R&C DM and show why this is inadequate for capturing the facts.

Under the R&C DM approach, DM operations are input-to-output rules which are ordered and apply in serial. If we call such a rule computation K following Embick (2010), then the grammar will 'apply computation K when the structural description of K is met' (p 40) (also the principle of 'Rules Apply' - Bobaljik 2017). Under R&C DM, constraints may exist as inviolable filters which can trigger a rule to repair illicit structures, e.g. the constraint T-NONINITIALITY in Basque triggering a number of repairs discussed on pg. 3. Extended to Degema, we would state that the operations DNI and LD are repairs to satisfy an analogous filter.

There are two main arguments against this view. First, a conceptual one is the **duplication** argument, common in debates of rules versus constraints. This states that if constraints are necessary in a derivation, and if constraints can derive the patterns that rules can, then rules only serve to duplicate derivational mechanisms and are therefore superfluous. Second, an argument particular to the Degema data involves the inadequacy of rule ordering in accounting for the data. As most work in DM is rule/R&C based, it is critical to establish the order of operations. It is frequently claimed or assumed that feature copying/dissociated node insertion takes place before vocabulary insertion (VI), which takes place before local dislocation (Embick & Noyer 2001, 2007; Embick 2007b). The common evidence justifying this ordering involves bleeding and feeding relations, e.g. in the Latin =que example in (22) above, the phonological properties of the VI inserted after =que will affect its LD pattern.

(33) Embick & Noyer (2001) - DM order of operations DNI > VI > LD

This order of operations predicts the wrong patterns in Degema. Both the single-marking and the double-marking SVC patterns derive from the same syntactic structure, and therefore the only difference in their post-syntactic inputs is the absence vs. presence of an overt intervening DP object. This is shown in the tables below using the double-marking context $mi=diw=n \ o'yi \ mi=ta=an$ 'I went with her/him' (from (8) above), versus the single-marking Ohoso $\rho=ta \ O \ de=n \ isen$ 'Ohoso went and bought fish' with no intervening object (from (11) above). Table 6 illustrates that the standard rule order correctly predicts the attested surface form.

Syntax	$\begin{bmatrix} ASP^{\circ} & V_1^{\circ} + V_1^{\circ} & DP + V_2^{\circ} & V_2^{\circ} + V_2^{\circ} \end{bmatrix} \end{bmatrix}$
DNI	$(\operatorname{asp}^{\circ}) (\operatorname{agr}_{\operatorname{sbj}} + v_1^{\circ} + V_1^{\circ}) (DP) (\operatorname{agr}_{\operatorname{sbj}} + v_2^{\circ} + V_2^{\circ} + \operatorname{agr}_{\operatorname{asp}})$
VI	(n)*(mi+duw)*(oyi)*(mi+ta+n)
LD	(mi+duw+n)*(oyi)*(mi+ta+n)
Predicted:	mi=duw=n oyi mi=ta=n ③ Attested: mi=duw=n oyi mi=ta=n

Table 6: Correct prediction in double-marking context with intervening DP object

In contrast, the single-marking table illustrates that this rule order does not correctly predict the attested surface form. Here, because DNI applies first, both sets of clitics receive full agreement marking. This is then the input for VI which inserts proclitics and enclitics on both sets of verbs, followed by LD which dislocates the initial enclitic. The result is an incorrect form ${}^{*}\rho = ta = n \rho = de = n$.

constructions with respect to the distribution of clitics. If an auxiliary and verb are surface adjacent, then the proclitic falls only on the auxiliary (with the exception of obligatory double-marking expressing imperfective aspect, example (15) on p. 10).

However, the first four auxiliaries listed above nwanyki, manyki, manyki, ma_1 , and ku have unexpected tonal effects which I do not account for at present. For example, $m_i = m\dot{a}_1 duw wo t\dot{a}$ 'I would've followed you (but didn't)' with auxiliary ma_1 'would have' appears with a SVC and together bear a [HLH] tonal melody. I adopt a working hypothesis that auxiliaries are subject to the same grammar as SVCs involving local dislocation and dissociated node insertion, but trigger distinct tonal melodies compared to lexical verbs. Discussion of Degema auxiliaries is found on: Kari (2003b:40-50,121,170,209), Kari (2004:25,30,35-38,68,77,132,160-163,234,278,284-291,295,302,347), and Kari (2005b).

Syntax	$[_{ASPP} ASP^{\circ} [_{\nu P} v_1^{\circ} + V_1^{\circ}]$	$[\emptyset \Psi_1^{\circ} [_{\nu P} v_2^{\circ} +$	$V_2^{\circ} \left[\frac{V_2^{\circ}}{V_2^{\circ}} \right] \right]$
DNI	$(asp^{\circ})(agr_{sbj}+v_1^{\circ}+V_1^{\circ})$	$(agr_{sbj} + v_2^{\circ} + V)$	$(2^{\circ} + agr_{asp})$
VI	(n)*(o+ta)*(o+de+n)		
LD	(o+ta+n)(o+de+n)		
● [™] Predicted:	*o=ta=n o=de=n	⁽²⁾ Attested:	o=ta de=n

Table 7: Incorrect prediction in single-marking context with standard rule order

Here, because DNI and VI take place before LD, this bleeds the application of LD. Keeping standard DM rule order results in an **ordering paradox**.

Within an R&C DM model, the only adequate rule order is therefore VI > LD > DNI. Here, VI would feed LD given that an object pronoun undergoes LD and forms a constituent with the verb depending on whether it is prosodically light or heavy (recall from Table 3 p. 7 that prosodically light pronouns do not form a natural morphosyntactic class). Further, LD would bleed DNI given that adjacent verbs show single-marking rather than double-marking.

Syntax	$\left[_{ASPP} ASP^{\circ} \left[_{\nu P} v_1^{\circ}\right]\right]$	+V ₁ ° [Ø V ₄ ° [_{νI}	$v_2^{\circ}+$	$V_2^\circ [\underbrace{V_2^\circ}]]]]$	
VI	(n)*(tạ)*(<mark>d</mark> ẹ)				
LD	(tạ+ <u>d</u> ẹ+n)				
DNI	(o+ta+de+n)				
Predicted:	ọ=ta ḍẹ=n	③ Attested:		ọ=ta ḍẹ=n	
T 11 0 D		1	• . 1	1 177 1.0	D 1

Table 8: R&C Analysis - Correct prediction with order VI > LD > DNI

Even if we allow for parametric variation in DM rule order, there is a more serious issue with this analysis. Under R&C DM, constraints trigger repairs which manifest as DM operations. The constraint we have established for Degema is V=WF(INFL) which states that all verbs are well-formed with respect to verbal inflection. In Table 8 above, DNI can be understood as repairing this constraint by inserting the proclitic exponing subject phi-features. In contrast, it is not apparent how LD would be a repair for this constraint. In this table, the verbs and the factative enclitic =n undergo LD. At first glance we might say that the verbs undergo LD with each other in order to form a constituent marked with aspectual inflection, thereby satisfying V=WF(INFL). However, such LD between verbs takes place even in the absence of an enclitic. Recall that future tense/aspect is marked by a set 1 proclitic, H tone, and no enclitic, repeated below.

(34) moj=tá	(*mó́)= gḗn	énám
3sg.set1=go	(*3sg.set1=)look.a	it animal
'(s)he will go l	ook at an animal' [ohl	k_20170806]

If the verbs are adjacent, only the single-marking pattern is permitted, which would be derived via LD. However, at the stage that VI and LD apply, there are no overt clitics present. It would therefore not be possible to motivate LD as a repair for V=WF(INFL) as it takes place *prior* to DNI. This is illustrated below.

Syntax	$\left[_{ASPP} ASP^{\circ} \right]_{vP} v$	$_{1}^{\circ}+V_{1}^{\circ}\left[\oslash V_{1}^{\circ}\left[v_{2}^{\circ}+V_{2}^{\circ}\left[V_{2}^{\circ}\right]\right]\right]$
VI	Ø*(tạ)*(dẹ)	
LD	(tạ+ <mark>ḍẹ</mark>)	← Unclear motivation
DNI	(mo+ta+de)	
Attested:	mo=ta de	

Table 9: R&C Analysis - Unclear motivation for LD in the absence of inflectional exponents

Under a strict interpretation of V=WF(INFL), LD should not be licensed here and we therefore expect a doublemarking pattern, contrary to fact. LD would therefore apply in effect in preparation for DNI, a classic 'look-ahead' problem found in rule-based serial derivations. Although it is possible to attribute LD to being an independent rule unrelated to the constraint V=WF(INFL), this would fully undermine the morphological conspiracy itself which sought to unify verb LD and agreement DNI (see similar arguments laid out in Foley *to appear* for Georgian).

5. OT-DM: In support of a hybrid model

I account for the Degema morphological conspiracy within an OT-DM model. In the view espoused here, DM morphological operations DNI and LD are deconstructed into a series of constraints. The verb inflection requirement in Degema is enforced via a set of highly-ranked morphological markedness constraints requiring a particular output shape and type, which are ranked higher than other constraints. Markedness constraints are counterweighted by mapping constraints which enforce particular mappings between input and output structure, and other faithfulness and markedness constraints. This is schematized in Tableau 1 below, involving the three main input types: SVCs with no object / V₁ V₂ /, SVCs with a prosodically light object / V₁ D_{σ} V₂ /, and SVCs with a prosodically heavy object / V₁ D_{$\sigma\sigma$} V₂ /. These inputs all have the potential to map to verbal structures without any clitic marking ('None'), structures with single-marking ('Single'), and structures with double-marking ('Double').

I:		$/ V_1 V_2 /$		Mark Set 1	Map Set 1	Faith	Map Set 2
O ₁ :		None	$V_1 V_2$	1!			
O ₂ :	£	Single	$cl=[V_1 V_2]=cl$			1	4
O ₃ :		Double	$cl=[V_1]=cl cl=[V_2]=cl$			3!	2
I:		$/ V_1 D_{\sigma} V_2 /$		Mark Set 1	Map Set 1	Faith	Map Set 2
O ₁ :		None	$V_1 D_{\sigma} V_2$	1!			
O ₂ :	ų	Single	$cl=[V_1 D_{\sigma} V_2]=cl$			1	6
O ₃ :		Double	$cl=[V_1 D_{\sigma}]=cl cl=[V_2]=cl$			3!	4
I:		$/ V_1 D_{\sigma\sigma} V_2 /$		Mark Set 1	Map Set 1	Faith	Map Set 2
O ₁ :		None	$V_1 D_{\sigma\sigma} V_2$	1!			
O ₂ :		Single 1	$cl=[V_1 D_{\sigma\sigma} V_2]=cl$	1!		1	6
O ₃ :		Single 2	$cl=[V_1 V_2]=cl D_{\sigma\sigma}$		2!	1	4
O ₄ :	3	Double	$cl=[V_1]=cl D_{\sigma\sigma} cl=[V_2]=cl$			3	2

Tableau 1: Emergence of single and double-marking patterns (numbers = number of violations)

The no clitic marking pattern for all inputs is eliminated because it violates the highly ranked markedness constraint. Single-marking surfaces in the / $V_1 V_2$ / and / $V_1 D_{\sigma} V_2$ / contexts over double-marking. Here, neither the single-marking or double-marking candidate violates the highly ranked markedness or mapping constraints – Mark Set 1 and Map Set 1 – but the double-marking candidate incurs more violations of the Faith set than the single-marking candidate and is thus eliminated. In contrast, double-marking in the / $V_1 D_{\sigma\sigma} V_2$ / context surfaces over two potential single-marking outputs: $cl=[V_1 D_{\sigma\sigma} V_2]=cl \ with \ pronoun integration and <math>cl=[V_1 V_2]=cl \ D_{\sigma\sigma} \ with \ V$ dislocation over the pronoun. The first single-making candidate incurs a fatal violation of a constraint in Marked Set 1 and is thus eliminated, while the second single-marking candidate incurs a fatal violation from Map Set 1. For all derivations, the single-marking candidates have more violations of Map 2 constraints than the double-marking, but because Map 2 constraints are dominated they are rendered inert.

Non-formally, single-marking is preferred overall as it inserts the least number of dissociated nodes and therefore incurs the least markedness violations. However, a consequence of single-marking is local dislocation of V_2 . If this LD operation resulting from single-marking is too costly, then double-marking results as a secondary 'repair', ensuring verbs are marked with appropriate inflection. I formalize and exemplify this central insight below. See also Appendix 2, where I illustrate a **factorial typology** using the proposed constraint set.

5.1. The constraint set

The constraint set I adopt is in Table 10 on p. 21. These are split into three constraint-strata (C-Strata) separated by solid lines, within which are sets of constraints separated by a dotted line. C-Strata are crucially ordered and must appear in the order provided. The constraints within each C-Stratum are not crucially ordered, but grouped together based on their type. There are four different types of constraints. Faithfulness (Faith) and mapping (Map) constraints are evaluated by comparing the input structure to an output candidate. Markedness (Mark) and alignment (Align) constraints solely evaluate output candidates without reference to the input.

Within C-Stratum 1, the first constraint set Mark 1 is composed of a series of markedness constraints 1-7. Constraints 1-3 V=WF_MWD(AGR_{SBJ}), V=WF_MWD(asp), and V1>V2 enforce inflection on morphological words. Constraint 4 MWD=PRWD enforces the morphological word to be a well-formed prosodic word, ensuring that clitics are incorporated into a surrounding word. Constraints 5-7 are **Labeling Constraints**, which this paper introduces. Labeling constraints require that all morphological words be labeled with a category reflecting a prosodically strong

morpheme within that morphological word. For example, a syntactic verbal head which corresponds to a vocabulary item /V/ within a morphological word is labeled with $\{V\}$, as shown below.

(35)

a. $x^{\circ} <-> (/x/)_{\{x\}}$ b. $v^{\circ}+V^{\circ} <-> (/V/)_{\{V\}}$ c. $D^{\circ} <-> (/D/)_{\{D\}}$

The labeling constraint *COMPLEXMWD_{LABEL} prohibits having more than one morphological label on a single MWd, and $\{V\}>\{D\}$ prefers to have a label $\{V\}$ over $\{D\}$ when they are in competition. These will become important when we look at SVCs with objects.

Within C-Stratum 1, constraints 8-9 ALIGN-/agr_{sbj}/-L and ALIGN-/asp/-R are Align constraints and enforce alignment between the edge of a clitic and the edge of a morphological word. Constraint 10 LINEARITYMAP-IO:LEX is a mapping constraint, which evaluates the interface mapping between syntactic hierarchical structure (the input) and morphological structure (the output). This constraint evaluates whether the hierarchical order of lexical constituents in the input is reflected in the linear order of counterparts in the output, and can be parameterized for specific grammars or sub-grammars. For example, in Degema the transitive verb structure [$v^{\circ}+V^{\circ}$ [DP] ...] is always mapped to a linear order / V D / (head-initial order). This constraint can be more technically defined based on specific properties of syntactic heads and VIs. This is outside of the scope of this paper, and for our purposes we will assume that this constraint gives us a baseline expectation for syntax-to-morphology mapping.

Within C-Stratum 2, Faith and Mark 2 constraints counterweight the Mark 1 constraints in C-Stratum 1. Faith constraint 11 DEP-IO(NODE) penalizes the insertion of dissociated nodes such as subject and aspect agreement. Markedness constraints 12-13 assign a violation for every occurrence of subject or aspect agreement.

Finally, C-Stratum 3 contains two Map constraints. Constraint 14 MAP(WD_TYPE) states that a syntactic head x° which is not dominated by a head x° is mapped to a morphological word (MWd) (see Embick & Noyer 2001:574).¹² By default syntactic words should correspond to morphological words, e.g. both functional and lexical heads correspondence to MWds.¹³ Finally, constraint 15 LINEARITYMAP-IO:FNC limits manipulating the expected linear order of functional head exponents, e.g. the linear position of aspect with repect to lexical items.

¹² Embick & Noyer's (2001: 574) definition: 'at the input to Morphology, a node X° is (by definition) a *morphosyntactic word* (MWd) iff X° is the highest segment of an X° not contained in another X° '.

¹³ I specifically do not adopt Selkirk's (2011) MATCH[word, ω] constraint because this is formalized as matching syntactic constituents with phonological constituents directly, whereas I adopt that this is mediated by an initial morphological mapping.

Set	#	Constraint	Definition	Function
	1	V=WF_MWD(AGR _{SBJ})	For a MWd marked with label {Verb} ({V}), assign a violation if it is not marked with subject agreement	Verbal words have a proclitic
	2	V=WF_MWD(ASP)	For a MWd marked with label {Verb} ({V}), assign a violation if it is not marked with aspect	Verbal words have a enclitic
	3	V1>V2	Under competition, the first MWd marked with label {Verb} ({V}) (defined linearly) bears inflection over the second MWd marked {V}	Mark the first verb with inflection over the second when you can't mark both
Mark 1	4	MWD=PRWD	A morphological word is a well-formed prosodic word	Clitics and light object pronouns incorporate into a surrounding word
	5	$MWD_{\{LABEL\}}$	For a MWd, assign a violation if a prosodically strong morpheme of category M does not project a morphological label {M}	Words containing a verb are labeled {V}, prosodically strong pronouns/nouns as {D}, etc.
	6	$COMPLEXMWD_{\{LABEL\}}$	For a MWd, assign a violation if it is marked with more than one morphological label	Words with more than one lexical item are not labeled with both, i.e. *{V},{D}
	7	{V}>{D}	Under competition, a MWd should be marked with label {Verb} ({V}) over a label {D}	Mark words with a verb with a {V} label over a {D} label when you can't have both
Alian	8	ALIGN-/AGR _{SBJ} /-L	The left edge of an $/agr_{sbj}/$ morpheme coincides with the left edge of a MWd	Agreement proclitics appear first in the word
Angn	9	ALIGN-/ASP/-R	The right edge of an /asp/ morpheme coincides with the right edge of a MWd	Aspect enclitics appear last in the word
Map 1	10	LINEARITYMAP-IO:LEX	The hierarchical order of lexical constituents x° and y° in the input is reflected in the linear order of counterparts /x/ and /y/ in the output	Limits manipulating the expected linear order of lexical exponents (e.g. Vs, Prons, Ns)
Faith	11	DEP-IO(NODE) ¹⁴	Morphemes in the output correspond to syntactic terminal heads in the input	Penalizes the insertion of dissociated nodes (e.g. Agr)
Marila 2	12	*AGR _{ASP}	Assign a violation for every instance of aspect agreement	Don't have aspect agreement
Mark 2	13	*AGR _{SBJ}	Assign a violation for every instance of subject agreement	Don't have subject agreement
	14	MAP(WD_TYPE)	Map a syntactic head x° which is not dominated by a head x° to a morphological word (MWd)	By default, syntactic words should correspond to morphological words
Map 2	15	LINEARITYMAP-IO:FNC	The hierarchical order of a functional head x° with respect to any head y° in the input is reflected in the linear order of counterparts /x/ and /y/ in the output	Limits manipulating the expected linear order of functional head exponents (e.g. aspect)

Table 10: Constraint set accounting for Degema clitics patterns (solid lines indicate distinct crucially ordered Constraint-Strata)

¹⁴ See also Wolf (2015:372) DEP-M(F) which penalizes the insertion of features not present in input.

5.2. The candidate set

This above constraint ranking will be used to evaluate a candidate set. Each syntactic input is mapped to a morphological output. The input-output mappings for the distribution of verbal clitics in Degema is provided in Table 11. These recapitulate the distribution of clitics: the single-marking pattern is found when there is no prosodically heavy object, and double-marking is found when there is. In the table below and throughout, forward slashes // indicate the input and backward slashes \\ indicate the output.

Туре		Syntactic input			Correct morphological output		
V	/	$[_{aspP} asp^{\circ} [_{v1P} v_1^{\circ} + V_1^{\circ} [\emptyset $	\rightarrow	/	$(/agr_{sbj}/+/V/+/asp/)_{V}$		\
VV	/	$[_{aspP} asp^{\circ} [_{\nu 1P} v_1^{\circ} + V_1^{\circ} [\emptyset + V_1^{\circ}] v_2P /$	\rightarrow	\	$(/agr_{sbj}/+/V1/+/V2/+/asp/)_{V}$		\
		$v_2^{\circ}+V_2^{\circ}[V_2^{\circ}]]$					
$V D_{\sigma}$	/	$[_{aspP} asp^{\circ} [_{v1P} v_1^{\circ} + V_1^{\circ} [DP + V_1^{\circ}]]]$ /	\rightarrow	\	$(/agr_{sbj}/+/V/+/D_{\sigma}/+/asp/)_{V}$		\
$V D_{\sigma} V$	/	$[_{aspP} asp^{\circ} [_{\nu 1P} v_1^{\circ} + V_1^{\circ} [DP + V_1^{\circ}]_{\nu 2P} /$	\rightarrow	\	$(/agr_{sbj}/+/V1/+/D_{\sigma}/+/V2/+/asp/)_{V}$		\
		$v_2^{\circ} + V_2^{\circ} \left[\underbrace{V_2^{\circ}} \right] \right] $					
$V D_{\sigma\sigma}$	/	$[_{aspP} asp^{\circ} [_{v_1P} v_1^{\circ} + V_1^{\circ} [DP + V_1^{\circ}]]]$ /	\rightarrow	\	$(/agr_{sbj}/+/V/+/asp/)_{V} * (/D_{\sigma\sigma}/)_{D}$		\
$V D_{\sigma\sigma} V$	/	$[_{aspP} asp^{\circ} [_{\nu 1P} v_1^{\circ} + V_1^{\circ} [DP _{+}^{\circ} [_{\nu 2P} /$	\rightarrow	\	$(/agr_{sbj}1/+/V1/+/asp/)_{V} * (/D_{\sigma\sigma}/)_{D}$	*	\
		$v_2^{\circ}+V_2^{\circ}[V_2^{\circ}]]$			$(/agr_{sbj}2/+/V2/+/agr_{asp}2/)_{V}$		

Table 11: Input-output mappings in Degema

Any constraint ranking proposed needs to successfully and uniformly adhere to these input-output mappings.

In order to assess the success of an OT-DM model, I generated a list of output candidates to be evaluated. These output candidates compete and the most optimal candidate is selected. These candidates were systematically generated along a series of dimensions.

(36) Dimensions along which output candidates were generated

- a. **IncD**: Did the verb and the pronominal object form one MWd or two? i.e. (/V/+/D/) vs. (/V/) * (/D/)
- b. IncV: In SVCs, did the verbs form one MWd or two?
 i.e. (/V1/+/V2/) vs. (/V1/) * (/V2/)
- c. IncAsp: Did the aspect marker and the verb form one MWd or two? i.e. (/asp/+/V/) vs. (/asp/) * (/V/)
- LinAsp: What is the linear position of the aspect marker?
 i.e. (/asp/+/V/) vs. (/V/+/asp/)
- e. **SbjAgr**: Did the verb appear with subject agreement in the same MWd? i.e. (/V/) vs. $(/agr_{sbj}/+/V/)$
- f. **AspAgr**: Did the verb appear with aspect marking in the same MWd? i.e. (/V/) vs. $(/V/+/asp/)\sim(/V/+/agr_{asp}/)$
- g. **Label**: Does the label of the MWd reflect a prosodically strong morpheme within it? i.e. $(/V/+/D_{\sigma})_{\{V\}}$ vs. $(/V/+/D_{\sigma})_{\{D\}}$

Each dimension had a number of values, e.g. IncV was a simple Y/N whereas AspAgr is more complicated in that it was evaluated twice in SVCs, once for each verb (N/YV1/YV2/YBoth). Every value of every dimension was combined to produce a set of candidates. Simplex inputs such as the /V/ type ($[aspP asp^{\circ} [_{\nu 1P} \nu_1^{\circ} + V_1^{\circ} [\emptyset \vee_{+}^{\circ}]]$) had fewer output candidates (n=14) because not all dimensions or values were relevant. Complex inputs such as /V $D_{\sigma\sigma}$ V/ had many more output candidates (n=214).

The first three dimensions IncD, IncV, and IncAsp reflect the presence or absence of one type of change associated with the DM operation Local Dislocation: the conversion of an independent morphological word MWd to a subword (SWd), a shift in morphological 'typing' (see Embick 2007b). For example, separate verbs would be $(V)_{MWd} * (V)_{MWd}$ whereas morphologically-incorporated verbs would be $((V)_{SWd} + (V)_{SWd})_{MWd}$. The fourth dimension LinAsp reflects another aspect of Local Dislocation, namely whether the aspect marker appears in its expected position or appears 'dislocated' with respect to a surrounding element (what 'expected' refers to is discussed below). The dimensions SbjAgr and AspAgr are equivalent to the DM operation DNI.

The generation of competing output candidates makes this OT-DM analysis distinct from DM analyses not implemented in a constraint-based model. In this OT-DM analysis, DM operations LD and DNI are freely available in all derivations, and are not strictly licensed by a lexical item or a particular grammatical or phonological context.

Strictly speaking, all morphemes within a specific input-output mapping are potentially subject to dislocating or the insertion of morphological nodes. Only those candidates appearing with these changes which are optimizing are selected.

5.3. The constraint ranking and evaluation

The constraint ranking from Table 10 above is summarized below, dividing constraints into their C-Stratum/type.



This constraint ranking was determined using OTSoft v2.5 (Hayes et. al 2013) by providing the program with fully evaluated tableaux for the 6 contexts. I illustrate this constraint ranking in the tableaux below.

The first tableau involves the /V/ input which involves no other verbs and no objects, shown in Tableau 2. The winning output is candidate 1. Candidates 8-14 do not appear with subject agreement / agr_{sbj} / in the MWd containing the verb and therefore incur a fatal violation of V=WF_MWD(AGR_{SBJ}) are eliminated. Similarly candidate 7 violates V=WF_MWD(ASP) and is eliminated. In candidate 6, the asp° head is mapped to a VI /asp/ which appears in an expected pre-verbal position. However, because aspect clitics are prosodically deficient, if they appear within their own MWd they violate MWD=PRWD; because candidate 6 violates this it is eliminated. Candidate 5 incorporates /asp/ within the MWd containing the verb and therefore avoids a violation of MWD=PRWD; however this candidate is marked with two morphological labels {V} and {Asp} and therefore violates *COMPLEXMWD_{LABEL} and is eliminated. Finally, candidates 2-4 containing MWds which are solely marked with {V} but the /asp/ marker does not appear at the right edge of the MWd, thereby violating ALIGN-/ASP/-R.

The winning candidate is therefore candidate $1 \setminus (/agr_{sbj}/+/V/+/asp/)_{\{V\}} \setminus$ which does not incur any of these violations of constraint stratum (CS) 1. This candidate does incur lower ranking violations in CS 2 and 3, though they are non-fatal as all other candidates have been eliminated. This candidate appears with $/agr_{sbj}/$, violating both DEP-IO(NODE) and $*AGR_{SBJ}$ penalizing the insertion of dissociated nodes not present in the input. Further, this candidate violates mapping constraints MAP(WD_TYPE) because by default the Asp° head should map to an independent MWd which it does not, and LINEARITYMAP-IO:FNC because by default the Asp° head should be linearized before the verb.

			Constraint stratum (CS) 1 CS 2													CS 3		
Іпрі	ıt:	$\left[_{aspP} asp^{\circ} \left[_{v1P} v_{1}^{\circ} + V_{1}^{\circ} \left[\oslash \Psi_{+}^{\circ} \right] \right] \right]$	V=WF_MWD(AGR _{SBJ})	V=WF_MWD(ASP)	V1>V2	MWD=PRWD	MWD _{{LABEL} }	*COMPLEXMWD _{{LABEL} }	{V}>{D}	ALIGN-/AGR _{SBJ} /-L	ALIGN-/ASP/-R	LINEARITY MAP-IO:LEX	DEP-IO(NODE)	$^{*} m AGR_{ASP}$	$^{*} m AGR_{SBJ}$	MAP(WD_TYPE)	LINEARITYMAP-IO:FNC	
1	ß	$(/agr_{sbj}/+/V/+/asp/)_{V}$											1		1	1	1	
2		$(/agr_{sbj}/+/asp/+/V/)_{V}$									1!		1		1	1		
3		$(/agr_{sbj}/+/V/+/asp/+/agr_{asp}/)_{V}$									1!		2		1	1	1	
4		$(/agr_{sbj}/+/asp/+/V/+/agr_{asp}/)_{V}$									1!		2	1	1	1		
5		$(/agr_{sbj}/+/V/+/asp/)_{V}, {Asp}$						1!					1		1	1	1	
6		$(/asp/)_{Asp} * (/agr_{sbj}/+/V/+/agr_{asp}/)_{V}$				1!							2	1	1			
7		$(/asp/)_{\{Asp\}} * (/agr_{sbj}/+/V/)_{\{V\}}$		1!		1							1		1			
8		$(/V/+/asp/)_{V}$	1!													1	1	
9		$(/asp/+/V/)_{V}$	1!								1					1		
10		$(/asp/+/V/+/agr_{asp}/)_{V}$	1!								1		1	1		1		
11		$(/V/+/asp/+/agr_{asp}/)_{V}$	1!								1		1	1		1	1	
12		$(/V/+/asp/)_{V}, {Asp}$	1!					1								1	1	
13		$(/asp/)_{Asp} * (/V/+/agr_{asp}/)_{V}$	1!			1							1	1				
14		$(/asp/)_{\{Asp\}} * (/V/)_{\{V\}}$	1!	1		1												

Tableau 2: /V/ input type

Moreover, Tableau 3 and Tableau 4 illustrate how this constraint ranking accounts for the correct outputs with /V D_{σ} / and /V $D_{\sigma\sigma}$ / input types. Tableau 3 involving the /V D_{σ} / type is a condensed version of the tableau found in Appendix 1: Evaluation (full tableaux). Output candidates 16-31 do not appear with subject agreement, and candidates 14-15 do not appear marked by aspect, and are therefore eliminated. Constraints 8-13 are eliminated because they violate MWD=PRWD, either because the prosodically deficient aspect VI /asp/ is not incorporated into a surrounding MWd or because the prosodically deficient monosyllabic object pronoun is not incorporated. Candidate 7 (/agr_{sbj}/+/V/+/D_{σ}/+/asp/)_{{V},{D},{Asp}} involves incorporation of both /asp/ and /D_{σ}/ but is eliminated because it contains more than one morphological label violating *COMPLEXMWD_{LABEL}. Finally candidates 2-6 are eliminated as they contain an aspect marker which is not right-aligned to the MWd. As with the /V/ type above, the winning candidate here violates lower ranked constraints in CS 2 and 3. It incurs two violations of MAP(WD_TYPE) because both /D_{σ}/ and /asp/ do not belong to separate MWds.

			Constraint stratum (CS) 1 CS 2											CS 3			
Іпрі	ıt:	$\left[_{aspP} asp^{\circ} \left[_{\nu_{1P}} \nu_{1}^{\circ} + V_{1}^{\circ} \left[DP V_{4}^{\circ} \right] \right] \right]$	V=WF_MWD(AGR _{SBJ})	V=WF_MWD(ASP)	V1>V2	MWD=PRWD	MWD _{{LABEL} }	*COMPLEXMWD _{LABEL}	{V}>{D}	ALIGN-/AGR _{SB/} -L	Align-/asp/-R	LINEARITYMAP-IO:LEX	DEP-IO(NODE)	$^{*} m AGR_{ASP}$	*AGR _{SBJ}	MAP(WD_TYPE)	LINEARITYMAP-IO:FNC
1	6	$(/agr_{sbi}/+/V/+/D_{\sigma}/+/asp/)_{V}$											1		1	2	2
2		$(/agr_{sbi}/+/asp/+/V/+/D_{g}/)_{V}$									1!		1		1	2	
3		$(/agr_{sbi}/+/V/+/asp/+/D_{\sigma}/)_{V}$									1!		1		1	2	1
4		$(/agr_{sbi}/+/asp/+/V/+/D_{\sigma}/+/agr_{asp}/)_{V}$									1!		2	1	1	2	
5		$(/agr_{sbi}/+/V/+/asp/+/D_{\sigma}/+/agr_{asp}/)_{V}$									1!		2	1	1	2	1
6		$(/agr_{sbi}/+/V/+/D_{\sigma}/+/agr_{asp}/)_{V}$									1!		2	1	1	2	2
7		$(/agr_{sbj}/+/V/+/D_{\sigma}/+/asp/)_{V}, {D}, {Asp}$						1!					1		1	2	2
8		$(/agr_{sbj}/+/V/+/asp/)_{V} * (/D_{\sigma}/)_{D}$				1!							1		1	1	1
9		$(/asp/)_{\{Asp\}} * (/agr_{sbj}/+/V/+/D_{\sigma}/+/agr_{asp}/)_{\{V\}}$				1!							2	1	1	1	
		[Cand 10-12]				1!											
13		$(/asp/)_{\{Asp\}} * (/agr_{sbj}/+/V/+/agr_{asp}/)_{\{V\}} * (/D_{\sigma}/)_{\{D\}}$				2!							2	1	1		
14		$(/asp/)_{Asp} * (/agr_{sbj}/+/V/+/D_{\sigma}/)_{V}$		1!		1							1		1	1	
15		$(/asp/)_{Asp} * (/agr_{sbj}/+/V/)_{V} * (/D_{\sigma}/)_{D}$		1!		2							1		1		
16		$(/V/+/D_{\sigma}/+/asp/)_{V}$	1!													2	2
		[Cand 17-30]	1!														
31		$(/asp/)_{\{Asp\}} * (/V/)_{\{V\}} * (/D_{\sigma}/)_{\{D\}}$	1!	1		2											

Tableau 3: /V D_o/ input type

Further the condensed Tableau 4 illustrates the /V $D_{\sigma\sigma}$ / type with a heavy object. Output candidates 16-33 violate a V=WF_MWd constraint, and 14-15 violate MWD=PRWD. Candidates 5-13 all involve cases where both the /asp/marker and the heavy object / $D_{\sigma\sigma}$ / are incorporated into the MWd containing the verb. Because both /V/ and / $D_{\sigma\sigma}$ / are prosodically heavy, they each should project a morphological label, {V} and {D} respectively. Candidates 6-11 are only labeled with {V} while candidates 12-13 are only marked with {D}. Candidate 5 is marked with both {V} and {D}, but this violates the labeling markedness constraint *COMPLEXMWD_[LABEL]. Here, incorporation of the prosodically heavy / $D_{\sigma\sigma}$ / is too costly with respect to labeling constraints, and therefore appears in its own MWd. This constraint ranking therefore militates against 'unnecessary' local dislocation. Finally, as with the tableaux above, candidates 2-4 are eliminated as they contain an aspect marker which is not right-aligned to the MWd.

			Constraint stratum (CS) 1 CS 2								CS 3						
Inpu	ıt:	$\begin{bmatrix} asp^{\circ} & sp^{\circ} \end{bmatrix}_{v1}^{v1} v_{1}^{\circ} + V_{1}^{\circ} & DP + V_{1}^{\circ} \end{bmatrix}$	V=WF_MWD(AGR _{SBJ})	V=WF_MWD(ASP)	V1>V2	MWD=PRWD	MWD {LABEL}	*COMPLEXMWD {Label}	{V}>{D}	ALIGN-/AGR _{SBJ} /-L	ALIGN-/ASP/-R	LINEARITYMAP-IO:LEX	DEP-IO(NODE)	$^{*} m AGR_{ASP}$	*AGR _{SBJ}	MAP(WD_TYPE)	LINEARITYMAP-IO:FNC
1	ų	$(/agr_{sbi}/+/V/+/asp/)_{V} * (/D_{gg}/)_{D}$											1		1	1	1
2		$(/agr_{sbj}/+/asp/+/V/)_{V} * (/D_{\sigma\sigma}/)_{D}$									1!		1		1	1	
3		$(/agr_{sbj}/+/V/+/asp/+/agr_{asp}/)_{V} * (/D_{\sigma\sigma}/)_{D}$									1!		2		1	1	1
4		$(/agr_{sbj}/+/asp/+/V/+/agr_{asp}/)_{V} * (/D_{\sigma\sigma}/)_{D}$									1!		2	1	1	1	
5		$(/agr_{sbj}/+/V/+/D_{\sigma\sigma}/+/asp/)_{V}, {D}, {Asp}$						1!					1		1	2	2
6		$(/agr_{sbj}/+/V/+/D_{\sigma\sigma}/+/asp/)_{V}$					1!						1		1	2	2
7		$(/agr_{sbj}/+/asp/+/V/+/D_{\sigma\sigma'})_{V}$					1!				1		1		1	2	
8		$(/agr_{sbj}/+/V/+/asp/+/D_{\sigma\sigma'})_{V}$					1!				1		1		1	2	1
9		$(/agr_{sbj}/+/asp/+/V/+/D_{\sigma\sigma}/+/agr_{asp}/)_{V}$					1!				1		2	1	1	2	
10		$(/agr_{sbj}/+/V/+/asp/+/D_{\sigma\sigma}/+/agr_{asp}/)_{V}$					1!				1		2	1	1	2	1
11		$(/agr_{sbj}/+/V/+/D_{\sigma\sigma}/+/asp/+/agr_{asp}/)_{V}$					1!				1		2	1	1	2	2
12		$(/V/+/D_{\sigma\sigma}/+/asp/)_{D}$					1!		1							2	2
13		$(/agr_{sbj}/+/V/+/D_{\sigma\sigma}/+/asp/)_{D}$					1!		1				1		1	2	2
14		$(/asp/)_{\{Asp\}} * (/agr_{sbj}/+/V/+/agr_{asp}/)_{\{V\}} * (/D_{\sigma\sigma}/)_{\{D\}}$				1!							2	1	1		
15		$\frac{(/\operatorname{asp})_{\{\operatorname{Asp}\}}}{(/\operatorname{agr}_{\operatorname{sbj}}/+///+/D_{\operatorname{or}}/+/\operatorname{agr}_{\operatorname{asp}}/)_{\{V\}}}$				1!	1						2	1	1	1	
16		$(/asp/)_{\{Asp\}} * (/agr_{sbj}/+/V/)_{\{V\}} * (/D_{\sigma\sigma}/)_{\{D\}}$		1!		1	- 1						1		1	1	
17		$\frac{(/\operatorname{asp})_{\{\operatorname{Asp}\}}}{(/\operatorname{agr}_{\operatorname{sbj}}/+///+//D_{\sigma\sigma})_{\{V\}}}$	1.	1!		1	1						I		1	1	1
18		$(/V/+/asp/)_{V} * (/D_{\sigma\sigma}/)_{D}$	1!													1	1
		[Land 19-31]	1!	1		1											
32		$\frac{(/asp/)_{Asp}}{(/asp)} \stackrel{\text{r}}{\leftarrow} \frac{(/V/)_{V}}{(/D_{\sigma\sigma})_{D}}$	1!	1		1	1									1	
55		$(/asp/)_{\{Asp\}} = (/V/+/D_{\sigma\sigma}/)_{\{V\}}$	1 !	1		1	1									1	

Tableau 4: /V D_{oo}/ input type

From these tableaux, the effects of DM operations LD and DNI emerge under specific constraint rankings, shown below. Crucially, the two components of LD – namely dislocating of a morphological constituent and change the type of a morphological constituent – are split into two sets of constraint interactions, entailing that they are separate individual components.

(38) Emergence of DM operations LD and DNI

a. DNI: Insertion of morphological nodes not present in input

EV-WE MWd(asp)	Dep-IO(Node)
V = WF = MWd(asp) >>	*agr _{asp}
$[v = w F_{iv} w d(agr_{sbj})]$	*agr.

- b. LD (dislocating): Dislocating MWd/SbWd with respect to local MWd/SbWd [Align-/asp/-R] >> [LinearityMap-IO:Fnc]
- c. **LD (typing):** MWd to SbWd morphological type-shifting $\begin{bmatrix} MWd=PrWd \\ V=WF_MWd(asp) \end{bmatrix} >> [Map(Wd_Type)]$

I now turn to how this constraint ranking can derive the correct outputs in serial verb constructions. Each of these input-output mappings has considerably more output candidates, i.e. the /V V/ type (n=86), the /V D_{σ} V/ type (n=206), and the /V $D_{\sigma\sigma}$ V/ type (n=214). Again, these tableaux below are condensed versions of those in Appendix 1: Evaluation (full tableaux).

Tableau 5 involves the /V V/ input. Like in the tableaux above, candidates 30-39 and 40-86 are eliminated as they violate V=WF_MWD(AGR_{SBJ}) and V=WF_MWD(ASP) respectively. Further, candidates 23-29 violate MWD=PRWD by not incorporating /asp/, candidates 11-22 are eliminated because not all /agr_{sbj}/ markers align to the left edge, and 3-10 are eliminated because not all /asp/ markers align to the right. Unlike in the tableaux above, there are *two* remaining candidates 1-2 which do not violate any constraint in CS 1. In both candidates, MWds which contain verbs appear with appropriate inflection and obeys all alignment constraints. In candidate 2 (the loser), the verbs appear in separate MWds and each are marked with a full set of inflectional markers, incurring three violations of DEP-IO(NODE) (one for each agreement marker). In contrast, candidate 1 (the winner) involves the incorporation of /V2/ with /V1/ and /V2/ forming a single /MWd/, with only one violation of Dep-IO(Node). This winning form has additional violations of mapping constraints in CS 3, but they are non-fatal.

				Constraint stratum (CS) 1									CS 2			CS 3	
Inpu	ıt:	$\left[_{aspP} asp^{\circ} \left[_{\nu_{1}P} \nu_{1}^{\circ} + V_{1}^{\circ} \left[\oslash \Psi_{4}^{\circ} \left[_{\nu_{2}P} \nu_{2}^{\circ} + V_{2}^{\circ} \left[\Psi_{2}^{\circ} \right] \right] \right] \right]\right]$	V=WF_MWD(AGR _{SBJ})	V=WF_MWD(ASP)	V1>V2	MWD=PRWD	MWD {LABEL}	*COMPLEXMWD _{LABEL}	{V}>{D}	ALIGN-/AGR _{SBJ} /-L	ALIGN-/ASP/-R	LINEARITYMAP-IO:LEX	DEP-IO(NODE)	$^{\rm *AGR_{ASP}}$	$^{\rm *AGR_{SBJ}}$	MAP(WD_TYPE)	LINEARITYMAP-IO:FNC
1	ų	$(/agr_{sbj}1/+/V1/+/V2/+/asp/)_{V}$											1		1	2	2
2		$(/agr_{sbj}1/+/V1/+/asp/)_{V} * (/agr_{sbj}2/+/V2/+/agr_{asp}1/)_{V}$											3!	1	2	1	1
3		$(/agr_{sbj}1/+/asp/+/V1/+/V2/)_{V}$									1!		1		1	2	
		[Cand 4-9]									2!						
10		$(/agr_{sbj}1/+/V1/+/agr_{asp}1/+/V2/+/asp/+/agr_{asp}2/)_{V}$									2!		3	2	1	2	2
11		$(/V1/+/agr_{sbj}1/+/V2/+/asp/)_{V}$								1!			1		1	2	2
		[Cand 12-21]								1!							
22		$(/agr_{sbj}1/+/V1/+/agr_{asp}1/+/agr_{sbj}2/+/V2/+/asp/+/agr_{asp}2/)_{V}$								1!	2		4	2	2	2	2
23		$(/asp/)_{Asp} * (/agr_{sbj}1/+/V1/+/V2/+/agr_{asp}1/)_{V}$				1!							2	1	1	1	
		[Cand 24-28]				1!											
29		$(/asp/)_{Asp} * (/agr_{sbj}1/+/V1/+/agr_{asp}1/+/agr_{sbj}2/+/V2/+/agr_{asp}2/)_{V}$				1!				1	1		4	2	2	1	
30		$(/agr_{sbj}1/+/V1/+/asp/)_{V} * (/agr_{sbj}2/+/V2/)_{V}$		1!									2		2	1	1
		[Cand 31-39]		1!													
39		$(/asp/)_{\{Asp\}} * (/agr_{sbj}1/+/V1/)_{\{V\}} * (/agr_{sbj}2/+/V2/)_{\{V\}}$		2!		1							2		2		
40		$(/V1/+/V2/+/asp/)_{V}$	1!													2	2
41		$(/agr_{sbi}1/+/V1/+/asp/)_{V} * (/V2/+/agr_{asp}1/)_{V}$	1!										2	1	1	1	1
		[Cand 42-85]	1!														
86		$(/asp/)_{\{Asp\}} * (/V1/)_{\{V\}} * (/V2/)_{\{V\}}$	2!	2		1											

Tableau 5: /V V/ input type (Condensed tableau)

Tableau 6 (p. 31) presents a condensed tableau with the /V D_{σ} V/ input type and Tableau 7 (p. 32) a condensed tableau with the /V $D_{\sigma\sigma}$ V/ type. In both tableaux, a large number of candidates are eliminated by incurring a fatal violation of highly ranked V=WF_MWd(agr_{sbj}), V=WF_MWd(asp), or MWd=PrWd. Of crucial importance to note, in Tableau 7 candidates 26-62 are eliminated as they have output structures involving /V/ and / $D_{\sigma\sigma}$ / within the a single MWd (an incorporated pronoun) but are marked by only one morphological label, parallel to the candidates eliminated in Tableau 4 with the /V $D_{\sigma\sigma}$ / input type. This ensures that a prosodically heavy pronoun forms its own MWd which it labels with {D} as in candidates 1-22.

In both tableaux, all other remaining candidates other than the top two are eliminated by violating an alignment constraint. The remaining two candidates in the /V D_{σ} V/ type in Tableau 6 are differentiated as they are in the /V V/ type: candidate 2 incurs more violations of Dep-IO(Node) than candidate 1. Compare this to the two remaining candidates in the /V $D_{\sigma\sigma}$ V/ type in Tableau 7. Here, although candidate $1 \left(/agr_{sbj}1/+/V1/+/asp/ \right)_{V} * (/D_{\sigma\sigma}/)_{D} * (D_{\sigma\sigma}/)_{D}$ more violations of Dep-IO(Node) than candidate $/agr_{sbj}2/+/V2/+/agr_{asp}1/)_{V} \land incurs$ 2 $(/agr_{sbj}2/\tau/V2/\tau/agr_{asp}1)_{V} * (/D_{\sigma\sigma}/)_{D} \setminus$ this latter candidate loses because it incurs more fatal violations of LINEARITYMAP-IO:LEX. Recall that this constraint states that the 'hierarchical order of lexical constituents x° and y° in the input is reflected in the linear order of counterparts x/a and y/a in the output', and thereby limits manipulating the expected linear order of lexical exponents such as verbs and nouns/pronouns. In Tableau 7, candidate 2 violates this constraint twice as $\frac{V2}{i}$ is not linearized after the object $\frac{D}{a}$ as expected, but rather incorporates over this element into the MWd containing /V1/. There are several alternative constraints one could posit - e.g. one involving a flavor of locality – but the result would be the same: this candidate is eliminated in favor of the double-marking pattern.

					(Constrai	int str	atum	(CS)	1				CS 2		C	S 3
Inpu	t:	$\left[_{aspP} asp^{\circ} \left[_{\nu 1P} \nu_{1}^{\circ} + V_{1}^{\circ} \left[DP \Psi_{+}^{\circ} \left[_{\nu 2P} \nu_{2}^{\circ} + V_{2}^{\circ} \left[\Psi_{2}^{\circ} \right] \right] \right] \right]$	V=WF_MWD(AGR _{SBJ})	V=WF_MWD(ASP)	V1>V2	MWD=PRWD	MWD _{{LABEL} }	*COMPLEXMWD _{{LABEL} }	{V}>{D}	ALIGN-/AGR _{SB/} /-L	ALJGN-/ASP/-R	LINEARITYMAP-IO:LEX	DEP-IO(NODE)	$^{*}AGR_{ASP}$	$^{\rm *AGR_{SBJ}}$	MAP(WD_TYPE)	LinearityMap-IO:Fnc
1	6	$(/agr_{sbj}1/+/V1/+/D_{\sigma}/+/V2/+/asp/)_{V}$											1		1	3	3
2		$(/agr_{sbi}1/+/V1/+/D_{\sigma}/+/asp/)_{V} * (/agr_{sbi}2/+/V2/+/agr_{asp}1/)_{V}$											3!	1	2	2	2
3		$(/agr_{sbi}1/+/asp/+/V1/+/D_{\sigma}/+/V2/)_{V}$									1!		1		1	3	
		[Cand 4-15]									1~2!						
16		$(/V1/+/D_{g}/+/agr_{sbi}1/+/V2/+/asp/)_{V}$								1!			1		1	3	3
		[Cand 17-33]								1!							
34		$(/agr_{sbi}1/+/V1/+/D_{g}/+/V2/+/asp/)_{V_{2}(D)_{2}(Asp)}$						1!					1		1	3	3
		[Cand 35-36]						1!									
37		$(/asp/)_{(Asp)} * (/agr_{sbi}1/+/V1/+/D_{g}/+/V2/+/agr_{asp}1/)_{V}$				1!							2	1	1	2	
		[Cand 38-71]				1~2!											
72		$(/asp/)_{Asp} * (/agr_{sbi}1/+/V1/+/agr_{asp}1/+/agr_{sbi}2/+/V2/+/agr_{asp}2/)_{V} * (/D_{\sigma}/)_{D}$				2!				1	1	2	4	2	2	1	
73		$(/agr_{sbi}1/+/V1/+/D_{\sigma}/+/asp/)_{V} * (/agr_{sbi}2/+/V2/)_{V}$		1!									2		2	2	2
		[Cand 74-94]		1!													
95		$(/asp/)_{Asp} * (/agr_{sbj}1/+/V1/)_{V} * (/D_{\sigma}/)_{D} * (/agr_{sbj}2/+/V2/)_{V}$		2!		2							2		2		
96		$(/V1/+/D_{\sigma}/+/V2/+/asp/)_{V}$	1!													3	3
		[Cand 97-205]	1~2!														
206		$(/asp/)_{\{Asp\}} * (/V1/)_{\{V\}} * (/D_{\sigma}/)_{\{D\}} * (/V2/)_{\{V\}}$	2!	2		2											

Tableau 6: /V D_{σ} V/ input type (Condensed tableau)

					С	onstra	aint stra	tum (CS) 1					CS 2		C	S 3
Input: $\begin{bmatrix} aspP & asp^{\circ} \end{bmatrix} \begin{bmatrix} v_1 P & v_1^{\circ} + V_1^{\circ} \end{bmatrix} DP = V_{+}^{\circ} \begin{bmatrix} v_2 P & v_2^{\circ} + V_2^{\circ} \end{bmatrix} V_{+}^{\circ} \end{bmatrix}$				V=WF_MWD(ASP)	V1>V2	MWD=PRWD	MWD _{{LABEL} }	*COMPLEXMWD _{{Label} }	{V}>{D}	ALIGN-/AGR _{SBI} /-L	ALJGN-/ASP/-R	LINEARITYMAP-IO:LEX	DEP-IO(NODE)	$^{\rm *AGR_{ASP}}$	*AGR _{SBJ}	MAP(WD_TYPE)	LINEARITYMAP-IO:FNC
1	ų	$(/agr_{sbj}1/+/V1/+/asp/)_{V} * (/D_{\sigma\sigma}/)_{D} * (/agr_{sbj}2/+/V2/+/agr_{asp}1/)_{V}$											3	1	2	1	1
2		$(/agr_{sbj}1/+/V1/+/V2/+/asp/)_{V} * (/D_{\sigma\sigma}/)_{D}$										2!	1		1	2	2
3		$(/agr_{sbj}1/+/asp/+/V1/)_{V} * (/D_{\sigma\sigma'})_{D} * (/agr_{sbj}2/+/V2/+/agr_{asp}1/)_{V}$									1!		3	1	2	1	
		[Cand 4-9]									1~2!						
10		$(/agr_{sbj}1/+/V1/+/agr_{asp}1/+/V2/+/asp/+/agr_{asp}2/)_{V} * (/D_{\sigma\sigma}/)_{D}$									2!	2	3	2	1	2	2
11		$(/V1/+/agr_{sbj}1/+/V2/+/asp/)_{V} * (/D_{\sigma\sigma}/)_{D}$								1!		2	1		1	2	2
		[Cand 12-21]								1!							
22		$(/agr_{sbj}1/+/V1/+/agr_{asp}1/+/agr_{sbj}2/+/V2/+/asp/+/agr_{asp}2/)_{V} * (D_{\sigma\sigma})_{D}$								1!	2	2	4	2	2	2	2
23		$(/agr_{sbj}1/+/V1/+/D_{\sigma\sigma}/+/V2/+/asp/)_{V}, {D}, {Asp}$						1!					1		1	3	3
		[Cand 24-25]						1!									
26		$(/agr_{sbj}1/+/V1/+/D_{\sigma\sigma}/+/V2/+/asp/)_{V}$					1!						1		1	3	3
		[Cand 27-61]					1~2!										
62		$(/agr_{sbj}1/+/V1/+/D_{\sigma\sigma}/+/agr_{sbj}2/+/V2/+/asp/)_{D}$					2!		1	1			2		2	3	3
63		$(/asp/)_{\{Asp\}} * (/agr_{sbj}1/+/V1/+/agr_{asp}1/)_{\{V\}} * (/D_{qq}/)_{(D)} * (/agr_{sbj}2/+/V2/+/agr_{asp}2/)_{(V)}$				1!							4	2	2		
		[Cand 64-75]				1!											
76		$(/asp/)_{(Asp)} * (/agr_{sbi}1/+/V1/+/D_{\sigma\sigma}/+/agr_{asp}1/+/agr_{sbi}2/+/V2/+/agr_{asp}2/)_{(V)}$				1!	1			1	1		4	2	2	2	
77		$(/agr_{sbi}1/+/V1/+/asp/)_{V} * (/D_{\sigma\sigma}/)_{D} * (/agr_{sbi}2/+/V2/)_{V}$		1!									2		2	1	1
		[Cand 78-100]		1~2!													
101		$(/asp/)_{Asp} * (/agr_{sbi}1/+/V1/+/D_{\sigma\sigma}/)_{V} * (/agr_{sbi}2/+/V2/)_{V}$		2!		1	1						2		2	1	
102		$(/agr_{sbj}1/+/V1/+/asp/)_{V} * (/D_{\sigma\sigma'})_{D} * (/V2/+/agr_{asp}1/)_{V}$	1!										2	1	1	1	1
		[Cand 103-213]	1~2 !														
214		$(/asp/)_{Asp} * (/V1/+/D_{\sigma\sigma}/)_{V} * (/V2/)_{V}$	2!	2		1	1									1	

Tableau 7: /V $D_{\sigma\sigma}$ V/ input type (Condenced tableau)

5.4. Morphological labeling

One innovation of this analysis is morphological labeling, e.g. MWds with verbs are marked with label {V}. Morphological labeling allows for sufficiently local verbs associated with independent V° heads to incorporate and form a single MWd, i.e. a morphological compound ($(/V1/)_{SbWd}+(/V2/)_{SbWd}$)_{MWd{V}}. Verbs which are not sufficiently local do not form a compound, e.g. the /V $D_{\sigma\sigma}$ V/ type does not become *($(/V1/)_{SbWd}+(/D_{\sigma\sigma}/)_{SbWd}+(/V2/)_{SbWd}$)_{MWd{V}} because of conflict with the incorporated /D/ labeling this MWd with a label {D}. In other words, morphological labeling facilitates the formation of a constituent involving only words of the same lexical category (e.g. V), and disallowing such a constituent if the words are of different categories (e.g. V and D). In terms of this analysis, morphological labeling counteracts potential 'mass local dislocation' for well-formedness conditions, wholesale incorporating of all morphemes into a single MWd to satisfy a markedness condition.

There is support in the literature that morphological labeling. Lexical categories such as nouns and verbs often display distinct phonological profiles even in the same language, e.g. in terms of the array of phonological contrasts, resistance to assimilation, among others (see references in Smith 2011). If these differences are analyzed as part of a speaker's synchronic grammar (rather than diachronic residue), then phonology requires access to lexical category information. For example, in Lenakel [tnl] (Oceanic: Vanuatu) primary stress falls on the penult (Lynch 1974, 1978). However, secondary stress in nouns falls on every other syllable from the primary-stress, whereas in verbs it is assigned from the initial syllable of the word (in both simplex and complex stems).

(39) Lenakel secondary stress

a. Nouns: secondary stress assigned rightward from primary-stress

	i. / nɨmʷakɨlakɨl /	$[ni.m^w \hat{\mathbf{p}}.ga.l\hat{\mathbf{a}}.gal]$ '	beach' (Lynch 1978:19)	
	ii. / kam-lomhanteni /	[kam.lɔ̀.m̥an.dɛ́.ni] '	for Lomhanteni' (Lynch 1974:83	5)
	iii. / kam-titoŋa /	[kàm .d ^y i. d 5.ŋa] '	for Titona' (Lynch 1974:183)	
b.	Verbs: secondary stress assi	gned leftward from init	ial syllable	
	i. / n-im-ausito /	[ni .maw. sí .do]	'you (sg.) told a story'	(Lynch 1974:66)
	ii. / n-im-ai-ausito /	[nì .ma.yu. sí .do]	'you (pl.) told a story'	(Lynch 1974:66)
	iii. / t-n-ak-am-ar-olkeik	ei / [tì .na. gà .ma.řəl.	gέy .gey] 'you (pl.) told a story'	(Lynch 1978:19)

Additionally, in Finnish there are vowel changes to stems when they appear adjacent to plural -i. With nouns, there is a preference for stem-final /a/ to change to /o/ in this context, but with adjectives in the same context there is a preference to delete stem-final /a/.

(40) Finish stem vowel alternation when adjacent to PL - i (Anttila 2002:13)

a.	Nouns: /a/ to /o	/ change preferred	
	/kihar a- i-ssa/	'curl-PL-INESSIVE'	[kihar o -i-ssa]
b.	Adjectives: dele	etion of /a/ preferred	
	/kihar a- i-ssa/	'curly-PL-INESSIVE'	[kiharØ-i-ssa]

Smith (2011) also notes English as an example (albeit a complicated one), where that is a 'a preference (not a requirement) for initial/trochaic stress in disyllabic nouns versus final/iambic stress in disyllabic verbs.'

It should be noted that cases where noun vs. verb phonology is carried over into derived words is rare in the literature, discussed in Inkelas (2014:14-15,50-59). Inkelas notes that 'it is much harder to find a language in which one phonological generalization holds of all nouns, whether monomorphemic or derived, and different phonological generalization holds of all verbs, whether monomorphemic or derived' (p. 51), and that 'even Japanese, perhaps the most-cited example in the literature of a noun-verb asymmetry because of its accentuation patterns, does not generalize the asymmetry to complex nouns and verbs' (p. 52).

Morphological labeling predicts to find morphological and phonological operations (clitic alignment requirements, phonological distributions or alternations) sensitive to the particular morphological label $\{M\}$, e.g. a language where tone spreads onto a MWd with label $\{V\}$ but not $\{N\}$ (controlling for relevant factors). This theory therefore makes testable predictions which should be evaluated with a larger typology of category-specific effects.

6. Against two alternatives

6.1. Alternative 1 – Syntactic verb movement

One alternative to my analysis is what I refer to as the *syntactic verb movement alternative*. Recall from the discussion of the syntax of Degema SVCs in section 3.3 (p. 9) that I assume a vP complementation structure of SVCs where V_1 selects v_2P as its complement: $[_{ASPP} ASP^{\circ} [_{vP} v_1^{\circ} [V_1^{\circ} [_{vP} v_2^{\circ} [V_2^{\circ}]]]]$ (Collins 1997, 2002). I have called verbs in the single-marking clitic pattern a *morphological compound* based on their acting as a single constituent with respect to clitic marking (and grammatical tone – section 4.3). Under my analysis, both single-marking and double-marking SVCs share the same syntactic structure and are subject to the same sequence of syntactic operations. In other words, both patterns are syntax-equivalent up to the point of spell-out.

In contrast, the syntactic verb movement alternative derives the constituency of the two verbs by overt syntactic head-movement, whether by V2 moving to V1 directly, or both V1 and V2 moving to the same higher functional head such as v1. Under this alternative, single-marking is attributed to the verbs forming a syntactic compound and therefore are inflected like non-compound verbs with a single set of clitics. Double-marking is attributed to the lack of syntactic verb movement under specific conditions resulting in the two verbs being spelled-out as separate words.

Syntactic verb movement in SVCs is argued for in Collins' (2002) analysis of [‡]Hoan verb compounds [huc] (Kxa, 'Khoisan': Botswana), exemplified in (41). Here, the verbs cluster in a pre-object field and are marked by a single aspect marker *a*-. Both arguments of the verbs appear afterwards.

(41) Verb compound derived via syntactic movement in #Hoan (Collins 2002:1)

ma	a- [q∥hu	' 0]	djo	ki	kx'u	na
1SG	PROG[pour	put.in]	water	PART	pot	in
	[V1	V2]	01		02	
(T			the met?	-				

'I am pouring water into the pot.'

I refer the reader to the original paper for syntactic details of this analysis.

This alternative is attractive in that it nullifies the need for local dislocation sensitive to particular lexical categories, and unifies the application of dissociated node insertion of agreement heads without stipulation. This syntactic alternative therefore provides us the best chance of success in accounting for the Degema patterns in a strictly serial rule-based DM model without constraints, optimization, and output-sensitivity.

6.1.1 Argument 1: Phonologically null objects show single-marking pattern

The first argument against the syntactic alternative is that phonologically null objects of V1 show the single-marking pattern. I demonstrated above that there is no semantic difference between the single- and double-marking pattern: all SVC types may show either pattern depending on the surface order of nouns and verbs, e.g. examples (7)-(8), p. 8. Consider the data in (42) involving question words in SVCs. When the question word *ovo* 'who' appears in-situ in object position (ex. a), the verbs are not sufficiently local and therefore cannot form a single MWd; as a consequence, each is marked with a set of clitics, the double-marking pattern. When the question word appears exsitu in a cleft construction, no object then intervenes between the verbs and they become sufficiently local and appear with single-marking (ex. b). It is ungrammatical in this ex situ context to mark the verbs with double-marking.

(42) Question word patterns in SVCs

a. In situ – Double-marking pattern mi=dúw=n óvo mi=tá=an ? 1SG.SET2=follow=FAC who 1SG.SET2=go=FAC 'I went with who?' (E.E. Kari p.c., 2015 Aug 24)

b. Ex situ – Single-marking pattern
ovó_i nú mi=dúw t_i tá=ān ? Cf. Ungram. *Ovó_i nú mi=dúūw t_i mi=tá=ān ?
who that 1SG.SET2=follow who go=FAC
'Who did I go with?' (E.E. Kari p.c., XXX) [ohk 2017XXXX]

Example (43) shows that single-marking likewise surfaces when an object is not present after V1 due to focus clefting (a.), relativization (b.), or object *pro*-drop (c.). This last example is especially telling as it involves pro-drop from two transitive verbs, resulting in five sufficiently local verbs. All five verbs form a MWd, with one proclitic on V1 and one enclitic on V5, and a H tone melody over the entire verbal complex.

(43) Single-marking pattern whenever verbs are sufficiently local

		1													
a.	Focus via c	lefting	g												
	kú óyi i		nų́	mi=q	ļúw			t _i		tá=ān					
	not her/	him	that	1SG.	set2=	follow	W	her/h	im	go=FAC					
	'It was not l	her/hi	im th	at I w	ent w	ith' (E	E.E. 1	Kari p.	c., 20	15.10.24)				
b.	Relativizati	on													
	owéy _i	nų́	mi=q	ļúw			<i>t</i> _i		tá=tē	[Cf. U	ngı	ram. *owéy _i i	nų́ mi=ợ	lúw= tē t₁	mi=tá=tē]
	person	that	1SG.	SET2=	follo	W	pers	son	go=P	RF					
	"the person	whon	n I ha	ve go	ne wi	th" (E	E.E. 1	Kari p.	c. 20	15.10.24)					
c.	Object drop	(indi	cated	by Ø)										
	Ohoso o=tá	L		dę́	Ø	vó	Ø	yį		kíyé=n	ó	byi			
	Ohoso 3sg.	SET2=	=go	buy	Ø	take	Ø	come		give=FA0	C h	ner/him			
	'Ohoso wer	nt and	boug	ht (so	methi	ng) a	nd b	rought	(it) to	her/him	.' (Kari 2004: 12	21)		

These data illustrate that clitic marking is insensitive to the selectional properties of the verbs, e.g. transitive vs. intransitive verb roots. If single-marking were the result of syntactic head movement, movement of the lower V2 head upwards would be triggered by a feature of a higher functional head. Under standard Minimalist assumptions, this predicts that by default when the syntactic structural condition is met, verb movement takes place. We therefore expect for single and double-marking patterns to be stable in the absence of an intervening object, as the presence of an object is orthogonal to the presence of a strong feature on a functional head. This is in particularly expected under a *Copy Theory of Movement* (Nunes 1995), where 'traces' are simply lower copies of moved constituents and are present in the syntax, but deleted at spell-out.

6.1.2 Argument 2: Unmotivated 'blocking' of head movement by an overt object

The second argument is that there is unmotivated 'blocking' of head movement by an overt object between V1 and V2, a counterpart to the first argument. Under the syntactic head movement alternative, V2 undergoes movement when triggered by a strong feature on a higher functional head. This movement therefore should be insensitive to whether there is an overt object present. Recall from the \ddagger Hoan verb compound example in (41) above that both verbs are subject to movement past their arguments, resulting in a [V1+V2 O1 O2] order (whether in complement or specific position). In general, in languages exhibiting verb compounding which (superficially) resemble the \ddagger Hoan type, the presence of an overt object does not block verb compounding (AUTHOR 201X: XX for languages/references).

We can compare this to the Degema facts. In Degema, V2 does not move over an intervening object to form a constituent with V1. For example, in (44) the transitive verbs tam 'chew' and dony 'swallow' appear in a SVC and share the internal object $idiy\bar{o}m$ 'food'. The object appears in its expected position between the verbs. It is ungrammatical to move V2 past the object, as in b... Similarly, (45) shows that verb movement is equally disallowed with a transitive V1P and an intransitive V2. Here, V1 and V2 cannot appear adjacent.

(44) No verb movement with two transitive verbs sharing internal argument

a.	Grammatica	Grammatical cl=V O cl=V										
	Jzakume	ó=tam	įdíyom	ọ= ḍóny								
	Jzakume	NEG\3SG.SET1=chew	food	3sg=swa	llow							
	'Jzakume d	id not chew food and	swallow' (Kari 2003	a: 278)							
b.	Ungrammat	tical *cl=V+ $V_i O t_i$										
	*Jzakume	ó=tam+ ḍọny i		įdíyom	t_i							
	Jzakume	NEG\3SG.SET1=chew	+swallow	food	swallow							
	Intended: 'J	zakume did not chew	food and s	swallow' (E.E. Kari p.c., XXX)							

(45) No verb movement with two transitive verb + intransitive verb (no argument sharing) Breno $o=d\hat{u}w$ mé tá=ān Cf. Ungram. *Breno $o=d\hat{u}w+t\hat{a}_i$ mé=ēn t_i Breno 3SG.SET1=follow me go=FAC 3SG.SET1=follow+go me=FAC go 'Breno went with me.' (Kari 2004:115; E.E. Kari p.c., XXX) It is unclear how the presence of an object in this specifier position could act as a syntactic blocker for headmovement, and weakens support for the syntactic movement alternative.¹⁵

6.1.3 Argument 3: SVCs with prosodically light pronouns show single-marking

A third argument against V2 head movement involves prosodically light pronouns. Recall that although objects generally block the single-marking SVC pattern, intervening prosodically light pronouns require it, i.e. the $cl=[V_1 D_{\sigma} V_2]=cl$ pattern. Under the syntactic movement alternative, because V1 and V2 form a single syntactic word via head-movement, and because D_{σ} still intervenes between them, it would be the case that D_{σ} also undergoes syntactic movement to this same complex head, a type of syntactic pronoun incorporation.

This is problematic for a number of reasons. First, recall that only prosodically light pronouns show incorporation; prosodically heavy pronouns condition the double-marking pattern. If under the syntactic movement analysis pronouns are also subject to syntactic incorporation, it is not clear how the syntactic trigger could only target prosodically light pronouns rather than all pronouns with an appropriate feature [D]. Under DM assumptions, syntax would not have access to phonological information before spell-out. Moreover, recall that the light pronouns do not form any natural class either with respect to their morphosyntactic features. Light pronouns expone feature bundles 1SG, 2SG, 2PL, and 3PL, while heavy pronouns expone 3SG and 1PL (see Table 3).

Further, monosyllabic pronouns and bisyllabic pronouns do not exhibit different syntactic behavior which would warrant different featural representations or syntactic positions. For example, prosodically light pronouns do not behave like syntactically incorporated pronouns, e.g. they are able to undergo clefting to express focus.

(46) Prosodically light pronouns can be clefted

a. Wóo u=món mé=ēn you 2SG.SET2=see me=FAC 'You saw me.' (Kari 2004: 164)
b. O=yí=n mé_i nú wo u=móōn t_i 3SG.SET2=be=FAC me that you 2SG.SET2=see\FAC me 'It was I who you saw' (Kari 2004: 164)

If the pronoun internally merges with a verb via head movement, and then subsequently moves out of that complex, this would be a type of 'excorporation' (Roberts 1991, 2011). There is very limited evidence for bona fide excorporation in the literature, with many arguing against the possibility of it (e.g. Julien 2002:67-87 and references therein; Matushansky 2006:95; a.o.).

6.2. Alternative 2 - Deletion-under-identity of intermediate clitics

Another alternative is what I term the *deletion-under-identity* (*DUI*) *alternative*. Under this alternative, the grammar uniformly generates a full set of proclitics and enclitics on all verbs in a SVC, but under specific conditions involving identity, there is obligatory deletion of intermediate clitics. This is assumed to be a type of **ellipsis**. This is schematized in (47) below.

(47) Alternative 2 - Deletion-under-identity of intermediate clitics

,				
		Single-marking	pattern	Double-marking pattern
a.	Uniform clitic marking	agr _{sbj} =V ₁ =asp Ø	agr _{sbj} =V ₂ =agr _{asp}	agr _{sbj} =V ₁ =asp DP agr _{sbj} =V ₂ =agr _{asp}
b.	Deletion-under-identity	agr _{sbj} =V ₁ =asp Ø	agr _{sbj} =V ₂ =agr _{asp}	$agr_{sbj} = V_1 = asp DP agr_{sbj} = V_2 = agr_{asp}$
c.	Surface pattern	$agr_{sbj} = V_1$	V ₂ =agr _{asp}	$agr_{sbj} = V_1 = asp DP agr_{sbj} = V_2 = agr_{asp}$

DUI would take place under two conditions: (1) the deleted clitics appear adjacent, and (2) the clitics be featurally identical. One advantage of this alternative is that Dissociated Node Insertion would take place uniformly on all verbs, and avoid any need to license Local Dislocation of verbs. This would therefore be compatible with a strictly serial rule-based analysis, and not warrant modifying core DM architecture. Such an alternative is flexible and could come in two flavors. The first would be that each verb's extended projection in the SVC contains a separate Asp° head; therefore the second aspect morpheme would not be agr_{asp} (see discussion in section 3.3). The second would

¹⁵ A limited pattern of verb compounding does exist in Degema with the verb *kiye* 'to give'. In a SVC, an allomorph of this verb *ke* appears right-adjacent to the other verb in the SVC, even if that other verb appears with an overt object. These patterns have not been analyzed at this time, and are superficially ambiguous between verbs in series vs. grammaticalization of *kiye* into a benefactive functional head *ke*.

be that there is one Asp° head, followed by concord aspect agreement. Both of these structures would result in the same input to a DUI operation.

There is precedence for similar types of DUI. Phenomena which fall under this rubric include coordination/conjunction reduction (Merchant 2012), 'suspended affixation' (Kabak 2007), and certain cases of verbal 'unbalanced coordination' (Johannessen 1998). An example of coordinate reduction in German nouns is in (48) below, where the medial morpheme *schaft* adjacent to the coordinator is deleted under identity.

(48) Coordinate reduction – German nouns

[[Freund] schaft] oder [[Feind] schaft]

'Friendship or hostility' (Booij 1985: 144)

6.2.1 Argument 1: The DUI alternative overgenerates

I present two arguments against a DUI alternative. The first is that this analysis makes the wrong predications and overgenerates with respect to a fuller set of Degema data. For example, one context which meets the surface conditions (adjacency and featural identicality) is covert coordination and other conjoined clauses. Covert coordination involves two clauses adjacent without a phonologically overt coordinator as in (49)a. The second clause can involve a linker auxiliary, e.g. *kiri* 'also.AUX' in (49)b. However, in these contexts DUI is not found and is in fact ungrammatical.

(49) Lack of DUI in conjoined clauses - Double-marking pattern obligatory

	5				01	U .	5
a.	$[V_1] \& [V_2]$						
	Ivioso o=kótú	mé=	ēn	•ekpérí=1	1	īnúm	
	Ivioso 3SG.SET2=call	me=	FAC	3SG.SET2	=tell=FAC	something	g
	'Ivioso called me and	told (me) s	omething'	(Kari 2003	a:274)	
b.	$[V_1]$ & $[aux V_2]$						
	Tatane o=kpé ēny		o =ki	rí	wậā	y	
	Tatane 3SG.SET2=wash	h\FAC	3SG.	SET2=also	.AUX sprea	ad\FAC	
	'Tatane washed and al	lso sp	read (something)'		
	cf. *o=kpény Ø kírí	í wáāy	y (E.E	. Kari p.c.,	2015 Oct 2	24) [ohk 2	017XXXX]
c.	$[V_1]$ but $[V_2]$	-		-			
	ó=kú	dí	bá a ∖	v dọ	o =rékérék	té	dí=īn
	3SG.SET2=did.AUX	eat	then	NFAC but	3SG.SET2	be.slow	eat=FAC
	'She did eat them but	she at	e the	n rather slo	owly'		
	cf. *ó=kú dí báaw do	Ø rék	éréké	dí=īn	-		
	cf. *ó=kú dí báw Ø do	9 0 =re	ékérék	té dí=īn			
	cf. *ó=kú dí báw Ø do	o Ø ré	kérék	té dí=īn	(E.E. Kari	i p.c., 2015	5 Dec 09)
				••	`	1 /	,

The following example comes from a Degema text (Kari 1997). In this context, two events are being described, one *moving* and one *settling*, each expressed in Degema through two SVCs in two clauses. Within each SVC, the verbs appear adjacent and therefore show single-marking. However, across the SVCs, because they appear in distinct clauses they cannot conflate into a one larger single-marking structure. Pronunciation of this sentence reveals that there is a clause obligatory pause between the final verb of SVC1 *desi* 'go far' and the first verb of SVC2 *ta* 'go' (E.E. Kari, p.c., 2016 May 12).

(50)

a. ...Banú Ipokuma, Obonogina o=vón=n éwéey nóonw o=wála pél Édá Sombreiro o=dá réré fún dési=īn o=tá jzá=n m'úlúgbó-éjzi útóm ísen gbódia, onāníná kúna, nu inám sáa

"...At Ipokuma, Obonogina left with his people and waded across the Sombreiro River, **moved and settled in the furthest part** of the wilderness for the purposes of fishing, farming, and hunting" (Kari 1997:64)

b. Single-marking within but not between SVCs

Under the DUI alternative, it is paradoxical why ellipsis does not take place across clauses as well as within them (cf. English *She_i* talked to me then $Ø_i$ left).

Further, under the DUI analysis the Degema single-marking pattern involves simultaneous backward deletion (deletion of material in the first conjunct) and forward deletion (deletion of material in the second conjunct) (Wilder 1995, 1997). Wilder (1995, 1997) illustrates that backward deletion and forward deletion are distinct operations subject to different phonological, syntactic, and semantic conditions, and we can therefore understand them as independent operations. However, in Degema when clitic deletion takes place, backward and forward deletion must take place simultaneously, and do not occur without each other. For example, in example (51) below, the proclitics share featural identity and the second proclitic appears at a conjunct boundary, identified as a common condition for DUI. However, deletion of the second proclitic is ungrammatical here and shows no variation for all speakers. The DUI alternative therefore overgenerates here.

(51)Ungrammatical forward deletion in the absence of backward deletion

a. Mi=dúw=n óyi *(mi)=tá=ān 1sG.SET2=follow=FAC her/him *(1sG.SET2)=go=FAC 'I went with her/him' (E.E. Kari p.c., 2015 Nov 02)
b. Tatane o=kótú=n óyi *(q)=kpérí=n īnúm Tatane 3sG.SET2=call=FAC him *(3sG.SET2)=tell=FAC something 'Tatane called him and told (him) something' (E.E. Kari p.c., 2015 Nov 02)

6.2.2 Argument 2: Requires DUI to be obligatory

One prevailing aspect of Deletion-Under-Identity and other types of ellipsis is that they are generally optional, noted overtly in Van Oirsouw (1985: 365). For example, the English sentence *he bought Ø and Ø cooked the chicken* can also surface as *he bought the chicken and he cooked the chicken*, if somewhat stilted-sounding. This is in stark contrast with the Degema clitic patterns where the single-marking and double-marking patterns are in complementary distribution. Ellipsis has been argued to be a 'repair' strategy in which case it is obligatory, e.g. Merchant (2001) on sluicing and repairing island violations and Kennedy & Merchant (2000) on N'-Ellipsis repairing Left Branch Condition violations. However, I do not see these cases of obligatory 'repair' ellipsis as relatable to the Degema facts presented here.

7. Discussion of post-syntactic architecture

As stated above summarizing the major tenets of DM (p.2), the major alteration which the OT-DM model makes is to the architecture of the syntax-phonology interface, viewing this transference as essentially parallelist and constraint-based, rather than serialist and rule-based. This is sketched below in Figure 2, adapted from Harley's (2014:228) DM schema and Broekhuis & Vogel's (2013:10) interface schema.



Figure 2: Revision of DM model according to OT-DM (PF branch only)

In this model, the numeration represents selecting a set of morphosyntactic features from the Feature Lexicon which are then subject to syntactic operations (e.g. MERGE, MOVE, AGREE, etc). At a certain point, this syntactic product is sent to spellout, which we can assume is triggered by a phase head (though this is orthogonal to the present

discussion). Under standard assumptions, spellout splits into two branches: a Phonetic/Phonological Form (PF) and a Logical Form (LF). The present alterations of the DM model affect the PF branch only.

At spellout, we can view the syntactic structure which is spelled out as a **Syntactic Image**. This image retains all syntactic constituency and dependencies such as c-command, but can no longer be manipulated by syntactic operations. This image acts as the Input to the morphology module in the Optimality Theoretic sense of the word. Morphology here can be understood as a series of morphological operations and adjustments which can affect the input image. These include: (1) **Bundle Manipulation** which alter the morphosyntactic featural content of input bundles (enrichment, impoverishment, fusion, fission), (2) **Bundle Mapping** (i.e. Vocabulary Insertion) which inserts Vocabulary Items from the Vocabulary into terminal nodes and where phonological content is supplied, (3) **Bundle Linearization**, which essentially translates the mobile-like hierarchical structure to a flat linear structure (linearization, local dislocation, prosodic inversion, etc.), and (4) **PF Interface Conditions** which enforce morphology specific requirements (e.g. dissociated node insertion, multiple copy resolution, a.o.).

As shown above with Degema, a candidate set generated from GEN is evaluated and the optimal candidate is selected. This in turn is fed into the Phonology module. OT-DM therefore unifies all of post-spellout on the PF branch as **Evaluative/Interpretive** via the same mechanisms: a ranked/weighted constraint set from CON, a list of output candidates from GEN, and an operation evaluating these candidates via EVAL. We can understand this as the **Evaluative** component of a derivation. This is in contrast with the **Generative** component of the derivation pre-spellout which does not involve these mechanisms.¹⁶

The strong claim of this proposal is that all operations take place in parallel and are not crucially ordered, which I call the **Morphology in Parallel Hypothesis (MPH).**

(52) Morphology in Parallel Hypothesis (MPH):

All morphological operations occur in parallel in a constraint-based model

MPH has a number of advantages. Given that Rules & Constraints DM à la Arregi & Nevins (2012) employs both rules and constraints, this results in a duplication problem which this hypothesis avoids. One of the original motivations of OT was eliminating duplication, as a constraint-based OT model accomplishes repairs and blocking and all other functions of a generative grammar (Prince & Smolensky 2004:239). Using the same architecture in both phonology and morphology is actually in the spirit of several practitioners of DM, such as Arregi & Nevins' (2012) design factor *Crossmodular Structural Parallelism* (see also discussion in Bobaljik 2017: section 3.4).¹⁷

As a final point of discussion, **morphological opacity** presents a significant challenge for the MPH, as opacity does for all parallelist models (Kiparsky 2000). **Counterfeeding opacity** refers to the underapplication of an operation when it would otherwise be expected, while **counterbleeding opacity** refers to the overapplication of an

See also Embick (2010:120) and discussion in Haugen (2011:10-13).

¹⁶ A stronger form of this model would be that the only thing which is truly Generative is the Numeration, and the evaluative component begins before spell-out. Such a view is explicitly argued for in the OT Syntax literature (e.g. Grimshaw 1997, a.o.; Legendre et al. 2001; Broekhuis & Vogel 2013; Legendre et al. 2016; a.o.). Under this conception, syntactic operations are also decomposed into OT constraints and input-output mappings are evaluated by EVAL. For example, principles such as the EPP are decomposed into a markedness constraint SUBJECT (Lengendre et al 2016;X).

 $^{^{17}}$ I have argued that a leading catalyst for shifting to a parallelist DM is the presence of morphological conspiracies, such as the one demonstrated by the Degema data. Underlying this motivation is an assumption that such conspiracies *should* be captured by a model of grammar. This is explicitly argued against by Embick (2010). He states that the incorporation of conspiracies into grammatical theory is due to a fear of the **Putative Loss of Generalization (PLG)**, which he views should rather be attributed to grammar-external factors:

[&]quot;An OT analysis can then say that the (phonological) grammar forces the attested distribution of allomorphs and, moreover, that the grammar explains the distribution by having morphological selection driven by optimization of the phonology of the output. The charge that is leveled against localist theories is that, while they might account for the distribution of allomorphs, they do not provide (within the grammar) a reason for the distribution. This type of argument against localist theories is based on their putative loss of generalization" Embick (2010:21)

[&]quot;To the extent that there are generalizations to be made about surface forms, the localist theory can make them, but they must be derivative of another part of language in the broad sense. That is, the explanations cannot be part of the grammar in the narrow sense; instead, they are the result of diachrony, acquisition, and so on." Embick (2010:21)

operation, e.g. when there is deletion of a conditioning environment. OT-DM can account for surface underapplication (counterfeeding type) by arguing that there is an additional higher markedness constraint which prevents the operation in a certain environment. Because R&C DM only involve constraints which are essentially filters, it cannot handle these cases because it would be assumed that such a filter would rule out the underapplication structure, contrary to fact. In contrast, R&C DM can account for the surface overapplication (counterbleeding type) as this is an expected result from rule-ordering and does not involve any filters. In contrast, OT-DM cannot account for this straightforwardly because the conditioning environment of the operation does not appear on the surface.

A purported case of morphological opacity from Arregi & Nevins (2012) involving Absolutive Promotion in Basque has been reanalyzed by Kiparsky (2017). He argues that all crucial ordering relations within the morphology are instances of feeding and bleeding, to which Arregi & Nevins (2017) respond with other claims of opacity in Basque morphotactics. I leave these issues aside, as the existence and status of morphological opacity is outside the scope of this paper, but note that it is fertile ground for testing the predictions of distinct DM models.¹⁸

8. Conclusion

This paper provided support for a modified DM model which I call Optimality Theoretic Distributed Morphology (OT-DM). The strongest form of this model was that all morphological operations take place in parallel, which I called the Morphology-in-Parallel Hypothesis (MPH). Although combining OT and DM is unorthodox in practice, I showed that a growing body warrant this modification, and provided evidence for OT-DM from a morphological conspiracy involving verbal clitics in Degema (Nigeria). To account for this conspiracy, I adopted that agreement clitics are inserted post-syntactically via the DM operation Dissociated Node Insertion (DNI), and further that verb complexes are formed post-syntactically via the operation Local Dislocation (LD). I argued that both these operations are triggered by a well-formedness markedness constraint which requires verbs to appear in properly inflected words on the surface. These DM operations were decomposed into a series of constraints which were crucially ranked. Candidates were freely generated from GEN and subject to all DM operations, and were evaluated via EVAL against the ranked constraint set. Finally, I illustrated that under standard views of DM in which DNI proceeds VI, a serial model results in an ordering paradox, and that even after parameterizing DM operation ordering in response, this model does not adequately account for the morphological conspiracy.

Gloss	Meaning	Abbreviation	Meaning
1	first person	cl	clitic
2	second person	D_{σ}	pronoun of as specific syllable count
3	third person	DNI	Dissociated Node Insertion
AUX	auxiliary	DP	Determiner Phrase
FAC	factative tense/aspect	LD	Local Dislocation
NEG	negative	MPH	Morphology-in-Parallel Hypothesis
NPM	non-past marker	OT-DM	Optimality Theoretic Distributed Morphology
PL	plural	R&C DM	Rules & Constraints Distributed Morphology
PRF	perfect aspect	SVC	serial verb construction
SET1	set 1 proclitic	V	verb
SET2	set 2 proclitic	VI	Vocabulary Insertion
SG	singular		

9. Glosses & abbreviations

10. Acknowledgments

This paper would not be possible without the expertise, insight, and generosity of collaborator Prof. Ethelbert E. Kari. Further thanks go to Ohoso Kari who checked the Degema data with me in summer 2017 in Port Harcourt,

(iii) yí dùw mè tá come follow me go 'come and follow me' (E E Kari

'come and follow me' (E.E. Kari p.c., 2015.08.24)

It is notable that in this context where there is no subject clitic, grammatical tone patterns are less predictable compared to the patterns presented in Section 4.3 (see also AUTHOR X:X).

¹⁸ In Degema, a potential case of counterfeeding opacity involving underapplication is the lack of agreement proclitics on verbs in positive singular imperatives, shown below. Compare positive plural imperatives and negative imperatives which *are* marked with a subject agreement proclitic (Kari 2004:34-38).

Nigeria. At Berkeley, thanks go to Peter Jenks, Line Mikkelsen, Larry Hyman, and Sharon Inkelas for reading drafts of this paper, and colleagues Nico Baier, Zach O'Hagan, Virginia Dawson, and Emily Clem. I am also thankful for discussions with Steven Foley, Jonathan Bobaljik, Ruth Kramer, and the audience of the 46th Annual Conference on African Linguistics at the University of Oregon.

11. References

Aboh, Enoch Oladé. 2009. Clause structure and verb series. Linguistic inquiry, 401. 1-33.

- Ameka, Felix K. 2005. Multiverb constructions on the West African littoral: Microvariation and areal typology. In: Mila Dimitrova-Vulchanova & Tor A. Åfarli (eds.) Grammar & beyond: essays in honour of Lars Hellan. 15-42. Oslo: Novus Press.
- Anderson, Stephen. R. 1992. A-morphous morphology. Cambridge University Press.
- Anttila, Arto. 2002. Morphologically conditioned phonological alternations. Natural Language & Linguistic Theory 20(1): 1-42.
- Arregi, Karlos & Andrew Nevins. 2017. Presyntactic morphology or postsyntactic morphology and explanatoriness in the Basque auxiliary. In Vera Gribanova & Stephanie S. Shih, The morphosyntax-phonology connection. NY:OUP.
- Arregi, Karlos & Andrew Nevins. 2012. Morphotactics: Basque auxiliaries and the structure of spellout. Vol. 86. Springer Science & Business Media.

AUTHOR. 2016.

- Baker, Mark & Osamuyimen T. Stewart. 2002. A serial verb construction without constructions. Unpublished Manuscript. Rutgers.
- Baker, Mark. 1989. Object sharing and projection in serial verb constructions. Linguistic Inquiry 20:513-553.
- Bermúdez-Otero, Ricardo & Kersti Börjars. 2006. Markedness in phonology and in syntax: the problem of grounding. Lingua 116 :710-756.
- Bobaljik, Jonathan. 2017. Distributed Morphology. Oxford Research Encyclopedia of Linguistics. Retrieved 6 Nov. 2017, from <u>http://linguistics.oxfordre.com/view/10.1093/acrefore/9780199384655.001.0001/acrefore-9780199384655-e-131</u>.
- Bonet, Eulàlia. 1994. The Person-Case constraint: A morphological approach. In Heidi Harley and Colin Phillips (eds.), MIT Working Papers in Linguistics 22: The Morphology-Syntax Connection. Cambridge, MA: MITWPL, pp. 33-52.
- Bonet, Eulàlia. 2013. Agreement in Two Steps (at Least). Distributed Morphology Today: Morphemes for Morris Halle: 167.
- Booij, Geert. 1985. Coordination reduction in complex words: A case for prosodic phonology. In (Eds.) H.v.d. Hulst and N. Smith, Advances in Nonlinear Phonology. Foris: Dordrecht. 143 160.
- Broekhuis, Hans & Ralf Vogel. 2013. Linguistic Derivations and Filtering. Minimalism and Optimality Theory. Advances in Optimality Theory.
- Brown, Jason. 2017. Non-adjacent reduplication requires spellout in parallel. Natural Language & Linguistic Theory, 35 (4), 955-977
- Caballero, Gabriela, & Sharon Inkelas. 2013. Word construction: Tracing an optimal path through the lexicon." Morphology 23.2: 103-143.
- Carstens, Vicki. 2002. Antisymmetry and word order in serial verb constructions. Language 78.1:3-50.
- Collins, Chris. 1997. Argument Sharing in Serial Verb Constructions. Linguistic Inquiry, 283:461-497.
- Collins, Chris. 2002. Multiple Verb Movement in Hoan. Linguistic Inquiry, 331:1-29.
- Dawson, Virginia. In press. Optimal clitic placement in Tiwa. In Proceedings of the Forty-Seventh Annual Meeting of the North East Linguistic Society, Vol. 1, ed. Andrew Lamont and Katerina A. Tetzloff, 243-256. Amherst: GLSA.
- Deal, Amy Rose. 2016. Plural exponence in the Nez Perce DP: a DM analysis. Morphology 26.3-4:313-339.
- Don, Jan, and Elma Blom. 2006. A constraint-based approach to morphological neutralization. Linguistics in the Netherlands 23.1: 78-88.
- Embick, David & Alec Marantz. 2008. Architecture and blocking. Linguistic Inquiry 39:1, 1 icv
- Embick, David & Rolf Noyer. 1999. Locality in post-syntactic operations Venue. In MITWPL 34: Papers on morphology and syntax, cycle two.
- Embick, David & Rolf Noyer. 2001. Movement operations after syntax. Linguistic Inquiry, 32(4). 555-595.
- Embick, David & Rolf Noyer. 2007. Distributed morphology and the syntax-morphology interface. Ramchand and Charles Reiss (eds.) Oxford handbook of linguistic interfaces.

- Embick, David. 2007a. Blocking effects and analytic/synthetic alternations. Natural Language and Linguistic Theory 25:1.
- Embick, David. 2007b. Linearization and local dislocation: Derivational mechanics and interactions. Linguistic analysis, 33(3-4), 2-35.
- Embick, David. 2010. Localism versus Globalism in Morphology and Phonology. MIT Press. Linguistic Inquiry Monographs.
- Foley, Steven. to appear. Morphological conspiracies in Georgian and Optimal Vocabulary Insertion. In the Proceedings of the 52nd Chicago Linguistics Society, eds. Jessica Kantarovich, Tran Truong, & Orest Xherija.
- Gribanova, Vera. 2015. Exponence and morphosyntactically triggered phonological processes in the Russian verbal complex. Journal of Linguistics, 51(3), 519-561. doi:10.1017/S0022226714000553
- Grimshaw, Jane. 1997. The best clitic: Constraint conflict in morphosyntax. Elements of grammar (1997): 169-196.
- Halle, Morris & Alec Marantz. 1994. Some key features of Distributed Morphology [MIT Working Papers in Linguistics 21: Papers on phonology and morphology], edited by A. Carnie & H. Harley, 275–288. Cambridge, Mass.: MITWPL.
- Halle, Morris. & Alec Marantz. 1993. Distributed morphology and the pieces of inflection. The View from Building 20, edited by K. Hale & S.J. Keyser, 111–176. Cambridge, Mass.: MIT Press.
- Halle, Morris. 1997. Distributed Morphology: Impoverishment and Fission. MIT Working Papers in Linguistics, 30, 425-449, 1997.
- Harley, Heidi & Rolf Noyer. 1999. Distributed Morphology. State-of-the-Article. Glot International 4(4). 3-9.
- Harley, Heidi. 2014. On the identity of roots. Theoretical Linguistics 2014; 40(3/4): 225 276
- Haugen, Jason. 2008. Morphology at the interfaces: reduplication and noun incorporation in Uto-Aztecan. John Benjamins Publishing Company (Linguistik Aktuell/Linguistics Today 117), Amsterdam/Philadelphia.
- Haugen, Jason. 2011. Reduplication in distributed morphology. In Coyote papers: Working papers in linguistics, Vol.
 18, eds. Jessamyn Schertz, Alan Hogue, Dane Bell, Dan Brenner, and Samantha Wray, 1–27. Tucson: University of Arizona Linguistics Circle
- Haugen, Siddiqi 2013. Roots and the Derivation. Linguistic Inquiry, Volume 44, Number 3, Summer 2013 493-517
- Hayes, Bruce, Bruce Tesar, & Kie Zuraw. 2013. OTSoft 2.5., software package, <u>http://www.linguistics.ucla.edu/people/hayes/otsoft/</u>.
- Hiraiwa, Ken & Adams Bodomo. 2008. Object-sharing as Symmetric Sharing: Predicate clefting and serial verbs in Dàgáárè. NLLT, 26:795-832.
- Inkelas, Sharon, and Cheryl Zoll. 2007. Is grammar dependence real? A comparison between cophonological and indexed constraint approaches to morphologically conditioned phonology. 133-171.
- Inkelas, Sharon. 2014. The interplay of morphology and phonology. Vol. 8. Oxford University Press.
- Johannessen, Janne Bondi. 1998. Coordination. Oxford: Oxford University Press.
- John J McCarthy. 2002. A Thematic Guide to Optimality Theory. Cambridge. Available at: http://works.bepress.com/john_j_mccarthy/93/
- John J McCarthy. 2008. Doing Optimality Theory. Malden, MA. Available at: http://works.bepress.com/john j mccarthy/95/
- Julien, Marit. 2002. Syntactic Heads and Word Formation. Cambridge University Press.
- Kabak, Bariş. 2007. Turkish suspended affixation. Linguistics 45-2 (2007), 311-347.
- Kager, Rene. 1999. Optimality Theory. Cambridge University Press.
- Kari, Ethelbert E. 1997. Degema. Vol. 180. Lincom Europa.
- Kari, Ethelbert E. 2002a. The source of Degema clitics. Languages of the world, 25. München: Lincom Europa.
- Kari, Ethelbert E. 2002b. Cliticization, movement, and second position. Languages of the world, 26. München: Lincom Europa.
- Kari, Ethelbert E. 2002c. Distinguishing between clitics and affixes in Degema, Nigeria. African study monographs, 233: 91-115.
- Kari, Ethelbert E. 2002d. Distinguishing between clitics and word in Degema, Nigeria. African study monographs, 234: 177-192.
- Kari, Ethelbert E. 2003a. Serial verb constructions in Degema, Nigeria. African Study Monographs, 244: 271-289.
- Kari, Ethelbert E. 2003b. Clitics in Degema: A meeting point of phonology, morphology, and syntax. Research institute for languages and cultures of Asia and Africa ILCAA, Tokyo. Dissertation.
- Kari, Ethelbert E. 2004. A reference grammar of Degema. Grammatische Analysen afrikanischer Sprachen. Vol 22. Köln: Rüdiger Köppe Verlag.
- Kari, Ethelbert E. 2005a. Degema subject markers: Are they prefixes or proclitics?. Journal of West African Languages, 321&2:13-20.

- Kari, Ethelbert E. 2005b. The grammar of Degema auxiliaries.In O.-M. Ndimele (ed.), Trends in the study of languages and linguistics: A festschrift for Philip Akujuoobi Nwachukwu, 499-509. Port Harcourt: Grand Orbit Communications and Emhai Press.
- Kari, Ethelbert E. 2006. Aspects of the Syntax of Cross-referencing Clitics in Degema. Journal of Asian and African Studies アジア・アフリカ言語文化研究 Ajia Afurika gengo bunka kenkyū, 72:27-38.
- Kari, Ethelbert E. 2008. Degema English dictionary with English index. Asian and African lexicon No. 52. Research institute for languages and cultures of Asia and Africa ILCAA. Tokyo University of Foreign Studies.
- Kari, Ethelbert E. 2015. Second position clitis in Degema: A reanalaysis;. A Paper Presented at the 3rd Department of African Languages & Literature International Conference held at the University of Botswana, Gaborone, Botswana, from 30 July 1 August 2014. Powerpoint presentation.
- Keine, Stefan & Gereon Müller. 2015. "Differential argument encoding by impoverishment." Scales and hierarchies. A cross-disciplinary perspective (2015): 75-130.
- Kennedy, Chris & Jason Merchant. 2000. Attributive comparative deletion. Natural Language & Linguistic Theory (2000) 18: 89. <u>https://doi.org/10.1023/A:1006362716348</u>
- Kinjo, Kunio, and Yohei Oseki. 2016. Wh-Concord in Okinawan= Syntactic Movement+ Morphological Merger." University of Pennsylvania Working Papers in Linguistics 22.1: 20.
- Kiparsky, Paul. 2000. Opacity and cyclicity. In Nancy A. Ritter (ed.), A review of Optimality Theory. Special issue, The Linguistic Review 17 (2-4): 351-67.
- Kiparsky, Paul. 2015. Stratal OT: A synopsis and FAQs. 2015. In Yuchau E. Hsiao and Lian-Hee Wee (eds.) Capturing Phonological Shades. Cambridge Scholars Publishing.
- Kiparsky, Paul. 2017. The morphology of the Basque auxiliary: Thoughts on Arregi & Nevins 2012. In Vera Gribanova & Stephanie S. Shih, The morphosyntax-phonology connection. NY:OUP.
- Kisseberth, Charles W. 1970. On the functional unity of phonological rules. Linguistic inquiry 1.3:291-306.
- Kisseberth, Charles W. 2011. Conspiracies. In Marc van Oostendorp, Colin J. Ewen, Elizabeth Hume and Keren Rice (eds.), The Blackwell Companion to Phonology.
- Kramer, Ruth. 2010. The Amharic Definite Marker and the Syntax-Morphology Interface. Syntax. 13(3). 196-240.
- Lahne, Antje. 2010. A multiple specifier approach to left-peripheral architecture. Linguistic Analysis 35:73-108.
- Legendre, G., Grismshaw, J., Vikner, S. (Eds.). 2001. Optimality-Theoretic Syntax. MIT Press, Cambridge, MA, pp. 1–27.
- Legendre, Gerladine. M. Putnam, H. de Swart, & E. Zaroukian (Eds.). 2016. Optimality-theoretic syntax, semantics, and pragmatics: From uni- to bidirectional optimization. Oxford: OUP.
- Lynch, John. 1974. Lenakel Phonology. Dissertation University of Hawaii. Working Papers in Linguistics [University of Hawaii], 7(1).
- Lynch, John. 1978. A grammar of Lenakel. Pacific Linguistics B55. Canberra: Australian National University.
- Martinović, Martina. 2017.. Nat Lang Linguist Theory 35: 205. https://doi.org/10.1007/s11049-016-9335-y
- Matushansky, Ora & Alec Marantz. 2013. Distributed Morphology Today : Morphemes for Morris Halle. Cambridge, US: MIT Press, 2014. ProQuest ebrary. Web. 12 June 2017. Copyright 017.017 MIT Press. All rights reserved.
- Matushansky, Ora. 2006. Head Movement in Linguistic Theory. Linguistic Inquiry. 37(1). 69-109.
- Merchant, Jason. 2001. The syntax of silence: Sluicing, islands, and the theory of ellipsis. Oxford University Press: Oxford.
- Merchant, Jason. 2012. Ellipsis. in: Syntax: An international handbook of contemporary syntactic research, ed. by Tibor Kiss and Artemis Alexiadou, Berlin: Walter de Gruyter. Draft Manuscript 2012, U Chicago.
- Moskal, Beata & Peter W. Smith. 2016. Towards a theory without adjacency: Hyper-contextual VI-rules. Morphology 26(3–4): 295–312.
- Moskal, Beata. 2015. Domains on the border: Between morphology and phonology. PhD diss., University of Connecticut.
- Newman, Stanley. 1944. The Yokuts Language of California. New York: The Viking Fund Publications in Anthropology.
- Norris, Mark. 2014. A theory of nominal concord. PhD Dissertation, UC Santa Cruz.
- Noyer, Rolf. 1992. Features, Positions and Affixes in Autonomous Morphological Structure. PhD thesis, MIT.
- Noyer, Rolf. 1993. Optimal Words: Towards a declarative theory of word-formation. Paper presented at ROW-1, New Brunswick, NJ.
- Noyer, Rolf. 1994. Mobile affixes in Huave: optimality and morphological well-formedness. In Proceedings of the Twelfth West Coast Conference on Formal Linguistics, edited by E. Duncan, D. Farkas, and P. Spaelti, 67-82. Stanford: CSLI.

- Nunes, Jairo 1995. The Copy Theory of Movement and Linearization of Chains in the Minimalist Program. PhD dissertation, University of Maryland, College Park.
- Offah, Kio Orunyanaa. 2000. A Grammar of Degema. Nigeria: Onyoma Research Publication.
- Opitz, Andreas. 2008. Case and markedness in Tlapanec. Linguistische Berichte 2008.214: 211-237.
- Paradis, Carole. 1987. On constraints and repair strategies. The linguistic review 6.1:71-97.
- Prince, Alan & Paul Smolensky. 2004 [1993]. Optimality Theory: Constraint Interaction in Generative Grammar. Rutgers University Center for Cognitive Science Technical Report 2.
- Roberts, Ian. 1991. Excorporation and minimality. Linguistic Inquiry, 22:209-218.
- Roberts, Ian. 2011. Head movement and the minimalist program. In Boeckx (ed.), The Oxford handbook of linguistic minimalism. Oxford: OUP. 195-219.
- Saab, A. and Lipták, A. 2016. Movement and deletion after syntax: Licensing by inflection reconsidered. Studia Linguist, 70: 66–108. doi:10.1111/stul.12039
- Salzmann, Martin. 2013. New arguments for verb cluster formation at PF and a right-branching VP: Evidence from verb doubling and cluster penetrability Source: Linguistic Variation [2211-6834] Salzmann, Martin yr:2013 vol:13 iss:1 pg:81
- Sande, Hannah. 2017. Distributing phonologically conditioned morphology. Dissertation, UC Berkeley.
- Selkirk, Elizabeth. 2011. The syntax-phonology interface. The handbook of phonological theory, 2, 435-483.
- Siddiqi, Daniel. 2009. Syntax within the Word: Economy, Allomorphy, and Argument Selection in Distributed Morphology. Linguistik Aktuell/ Linguistics Today.
- Siddiqi, Daniel. 2010. Distributed Morphology. Language and Linguistics Compass.
- Smith, Jennifer 2011. Category-specific effects. In Marc van Oostendorp, Colin Ewen, Beth Hume, and Keren Rice (eds), Companion to Phonology. Malden, MA: Wiley-Blackwell 2439-2463
- Stump, Gregory T. 2001. Inflectional morphology: A theory of paradigm structure. Vol. 93. Cambridge University Press.
- Trommer, Jochen. 2001b. A hybrid account of affix order. CLS 37: 469-480.
- Trommer, Jochen. 2002. Modularity in OT-morphosyntax. Resolving conflicts in grammars: Optimality Theory in syntax, morphology, and phonology. 83-117.
- Tucker, Matthew A. 2011. The morphosyntax of the Arabic verb: Toward a unified syntax-prosody. Linguistics Research Center (2011). Series: Morphology at Santa Cruz: Papers in Honor of Jorge Hankamer. 177-211. Permalink: <u>https://escholarship.org/uc/item/0wx0s7qw</u>
- van Driem, G. 1993. A Grammar of Dumi. Berlin and New York: Mouton de Gruyter.
- Van Oirsouw, Robert R. 1985. A linear approach to coordinate deletion. Linguistics 23. 363-390.
- Watanabe, Akira. 2015. Valuation as deletion: inverse in Jemez and Kiowa. Natural Language & Linguistic Theory 33(4): 1387-1420.
- Wilder, Chris. 1995. Rightward Movement as Leftward Deletion. In On Extraction and Extraposition in German, ed. Uli Lutz and Jürgen Pafel, 273-309. Amsterdam: John Benjamins.
- Wilder, Chris. 1997. Some properties of ellipsis in coordination. Alexiadou, Artemis, ed. Studies On Universal Grammar And Typological Variation. Amsterdam, NLD: John Benjamins Publishing Company. 59–107.
- Wolf, Matthew. 2008. Optimal Interleaving: Serial Phonology-Morphology Interaction in a Constraint-Based Model. Ph.D. dissertation, University of Massachusetts Amherst.
- Wolf, Matthew. 2015. Lexical insertion occurs in the phonological component. In Eulàlia Bonet, Maria-Rosa Lloret, and Joan Mascaró Altimiras (eds.), Understanding Allomorphy: Perspectives from Optimality Theory. Sheffield: Equinox, pp. 361-407.
- Xu, Zheng. 2016. The Role of Morphology in Optimality Theory. In A. Hippisley & G. Stump (Eds.), The Cambridge Handbook of Morphology (Cambridge Handbooks in Language and Linguistics, pp. 513-549). Cambridge: Cambridge University Press. doi:10.1017/9781139814720.020

12. Appendix 1: Evaluation (full tableaux)

[See separate files "FilesForappendix1_nllt" and "appendix 1 degema tableaux.xlsx"]

13. Appendix 2: Factorial typology of constraint set

[See separate files "FilesForappendix2_nllt_factyp" and "appendix 2 degema factorial typology.xlsx"]

Within Optimality Theory, a factorial typology refers to determining all the logical rankings of a set of constraints and computing the different winning sets of output candidates (Kager 1999:35). A factorial typology is particularly useful in determining if there are pathological predictions from a set of constraints. In the current data set, a potential example of a pathological prediction would be a grammar which inserted subject agreement when there is a

prosodically light object but not when there is a prosodically heavy one, a grammar which is not generated by the set of constraints proposed. By default, we expect the presence of subject agreement and the prosodic type of object to be orthogonal.

A factorial typology was determined using OTSoft v2.5 (Hayes et al. 2013). There were 15 constraints considered, making the logically possible number of grammars the factorial 15! (1307674368000). Restricting our inputs to the 6 input types presented above, the factorial typology resulted in 128 distinct grammars. The factorial typology does not reveal any pathological predictions, and therefore this constraint set is sufficiently restrictive. Major parameters of output variation include (1) not incorporating $/D_{\sigma}/$, (2) lacking subject agreement, (3) not incorporating /asp/ but having aspect agreement, (4) not dislocating /asp/ to the right edge, (5) dislocating /V2/ over a pronoun /D/, and (6) not incorporating /V2/ resulting in /V1/ and /V2/ forming separate MWds, with each verb receiving separate inflection i.e. the double-marking pattern.

A grammar of this last type is provided below with Grammar 18. Recall from section 3.2 that the /V V/ and /V D_{σ} V/ types with double-marking is interpreted as questionable ^(?) to ungrammatical * in this dialect of Degema depending on speaker. A grammar which optimizes double-marking here can be derived if we reorder constraint stratum 3 over constraint stratum 2, as shown below (C-Strata order established in Tableaux above).

Input type		Grammar 18 - /V2/ appears in its own MWd
/V/	\rightarrow	$(/agr_{sbj}/+/V/+/asp/)_{V}$
/V V/	\rightarrow	$(/agr_{sbj}1/+/V1/+/asp/)_{V} * (/agr_{sbj}2/+/V2/+/agr_{asp}1/)_{V}$
$/V D_{\sigma}/$	\rightarrow	$(/agr_{sbj}/+/V/+/D_{\sigma}/+/asp/)_{V}$
$/V D_{\sigma} V/$	\rightarrow	$(/agr_{sbj}1/+/V1/+/D_{\sigma}/+/asp/)_{V} * (/agr_{sbj}2/+/V2/+/agr_{asp}1/)_{V}$
$/V D_{\sigma\sigma}/$	\rightarrow	$(/agr_{sbj}/+/V/+/asp/)_{V} * (/D_{\sigma\sigma}/)_{D}$
$/V D_{\sigma\sigma} V/$	\rightarrow	$(/agr_{sbj}1/+/V1/+/asp/)_{V} * (/D_{\sigma\sigma}/)_{D} * (/agr_{sbj}2/+/V2/+/agr_{asp}1/)_{V}$
CS 3 Map(Wd_T Linearity-IC	ype) D:Fnc	$>> \begin{bmatrix} CS \ 2 \\ DEP-IO(Node) \\ *agr_{asp} \\ *agr_{shi} \end{bmatrix}$

Table 12: Constraint strata ranking and output candidate which corresponds to ?~*?~* grammaticality interpretation

A sample of grammars generated by the factorial typology is in Table 13.

Grammar	Label	Туре	Ι: /V D _σ V/	I: /V D _{σσ} V/	Ranking differences compared to attested (p. 23)
18	No V2 Inc	Don't incorporate /V2/	(sbj+V+D _o +asp) (sbj+V+asp)	$(sbj+V+asp) (D_{\sigma\sigma}) (sbj+V+asp)$	$\begin{bmatrix} 1 \\ Map(Wd_Type) \\ Linearity-IO:Fnc \end{bmatrix} \gg \begin{bmatrix} 2 \\ DEP-IO(Node) \\ *agr_{asp} \\ *agr_{sbi} \end{bmatrix}$
47	No Asp Inc	Don't incorporate /asp/ into /V/	$(asp) (sbj+V+D_{\sigma}+V)$	$(asp) (sbj+V) (D_{\sigma\sigma}) (sbj+V)$	$\begin{bmatrix} 1 \\ *agr_{asp} \\ LinearityMap-IO:Fnc \end{bmatrix} >> \begin{bmatrix} 2 \\ V=WF_MWd(asp) \\ MWd=PrWd \end{bmatrix}$ $>> \begin{bmatrix} 3 \\ Map(Wd \ Type) \end{bmatrix}$
15	V Over D	Incorporate /V2/ over /D/	$(sbj+V+V+asp) (D_{\sigma})$	$(sbj+V+V+asp) (D_{\sigma\sigma})$	$\begin{bmatrix} \frac{1}{*_{agr_{asp}}} \end{bmatrix} \gg \begin{bmatrix} \frac{2}{Map(Wd_Type)} \\ Lin-IO:Fnc \end{bmatrix} \gg \begin{bmatrix} \frac{3}{MWd} = PrWd \\ Lin-IO:Lex \end{bmatrix}$
74	Inc w/o Agr	V incorporation without agreement	$(asp) (V+D_{\sigma}+V)$	$(asp) (V) (D_{\sigma\sigma}) (V)$	$\begin{bmatrix} 1 \\ \text{Dep-IO(Node)} \\ * agr_{asp} \\ * agr_{sbj} \\ \text{Lin-IO:Fnc} \end{bmatrix} \Longrightarrow \begin{bmatrix} 2 \\ \text{V=MF}_{MWd(agr_{sbj})} \\ \text{V=MF}_{MWd(asp)} \\ \text{MWd=PrWd} \end{bmatrix}$
68	Cond Inc D	Incorporate D only if incorporate V2	(asp) (V+D _σ +V)	(asp) (V+D _{oo} +V)	$\begin{bmatrix} 1 \\ Dep-IO(Node) \\ *agr_{asp} \\ *agr_{sbj} \\ Lin-IO:Fnc \end{bmatrix} >> \begin{bmatrix} 2 \\ V=MF_MWd(agr_{sbj}) \\ V=MF_MWd(asp) \end{bmatrix}$ $>> \begin{bmatrix} 3 \\ MWd_{\{label\}} \\ Man(Wd_Type) \end{bmatrix} >> \begin{bmatrix} 4 \\ MWd=PrWd \end{bmatrix}$
7	D Label Repair	Bear label {D} to avoid agreement	$(sbj+V+D_{\sigma}+V+asp)_{\{V\}}$	$(V + D_{\sigma\sigma} + V + asp)_{\{D\}}$	$\begin{bmatrix} 1 \\ *_{agr_{asp}} \end{bmatrix} \gg \begin{bmatrix} 2 \\ Dep-IO(Node) \\ *_{agr_{sbi}} \end{bmatrix} \gg \begin{bmatrix} 3 \\ MWd_{\{label\}} \\ \{V\} > \{D\} \end{bmatrix}$
70	Multiple Labels	Distinct effects when have multiple labels	(asp) $(V+D_{\sigma}+V)_{\{V\}}$	$(V+D_{\sigma\sigma}+V+asp)_{\{V\}\{D\}\{ASP\}}$	$\begin{bmatrix} 1 \\ Dep-IO(Node) \\ *agr_{asp} \\ *agr_{sbj} \end{bmatrix} \implies \begin{bmatrix} 2 \\ V=MF_MWd(agr_{sbj}) \end{bmatrix}$ $\implies \begin{bmatrix} 3 \\ *ComplexMWd_{\{label\}} \\ Map(Wd_Type) \\ Lin-IO:Fnc \end{bmatrix} \implies \begin{bmatrix} 4 \\ V=WF_MWd(asp) \\ MWd=PrWd \end{bmatrix}$

Table 13

In Grammar 47 No Asp Inc, /asp/ is not incorporated into a surrounding verb and there is no aspect agreement. In Grammar 15 V Over D, /V2/ incorporates into the MWd containing /V1/ but the object /D/ is not incorporated. In this type, /V2/ actually incorporates over /D/ resulting in a change in word order. On the surface, therefore, such grammars may be ambiguous to speakers whether they involve this type of post-syntactic incorporation, or syntactic head movement like the one discussed above (Collins' analysis of $\frac{1}{2}$ Hoan verb compounds in Section 6.1).

Grammar 74 **Inc w/o Agr** involves the incorporation of /V2/ into /V1/ without the presence of agreement, i.e. an optimal output (asp) (V+D+V) (V+D+V) Such a structure results when constraints requiring verbal inflection (i.e. V=WF_MWD(agr_{sbj}) and V=WF_MWD(asp)) are ranked *below* constraints disallowing these morphemes (i.e. *agr_{sbj} and *agr_{asp}) but *above* a mapping constraint requiring terminal nodes map to MWds (i.e. Map(Wd_Type). In this case, incorporation takes place because it has less violations of the V=WF constraints.

Grammar 68 **Cond Inc D** involves incorporating /D/ only under the conditions that /V2/ is also incorporated, i.e. a non-SVC output $(asp) (V) (D_{\sigma\sigma}) but a SVC output <math>(asp) (V+D_{\sigma\sigma}+V) \\$. In this case, /D/ is incorporated only if it would otherwise 'get in the way' of V2 incorporating, and not otherwise. The constraint ranking which accounts for this involves ranking the V=WF_MWd constraints and Map(Wd_Type) crucially above MWd=PrWd.

The final two grammars involve morphological labeling, as discussed in Section 5.4. In grammar 7 **D** Label **Repair**, the optimal output $\langle (V+D+V+asp)_{\{D\}} \rangle$ bears a label $\{D\}$ rather than the label $\{V\}$ in order to avoid a violation for not bearing agreement. This output is found in grammars 7,14,16,27,29,36,43,45,50,57,59,64. Similarly, grammar 70 **Multiple Labels** involves effects when one of the candidates has multiple labels. In this case when the optimal output candidate has a $\{V\}$ label, the /asp/ marker surfaces appears in its own MWd, whereas in another derivation the optimal output has $\{V\}$, $\{D\}$, and $\{Asp\}$ labels and /asp/ appears at the right edge of this MWd. This output is found in grammars 39,46,53,60,70,77,84,91.

It is unclear at this point whether the last two types of grammars constitute pathological predictions of the constraint set. This is because although candidates were generated systematically along several important dimensions (described in Section 5.2), generating the complete list of candidates with different morphological labels was not feasible for this project. I therefore suspect therefore that these final two grammar types are merely an artefact of the limited number of candidates evaluated in this study with respect to different morphological labels.